IEEEAccess Multidisciplinary : Rapid Review : Open Access Journal

Received 13 February 2024, accepted 28 February 2024, date of publication 1 March 2024, date of current version 21 March 2024. Digital Object Identifier 10.1109/ACCESS.2024.3372852

RESEARCH ARTICLE

Dynamic Decision Making Process for Dangerous Good Transportation Using a Combination of TOPSIS and AHP Methods With Fuzzy Sets

HASSAN KANJ[©]¹, (Member, IEEE), YEHIA KOTB[©]¹, (Member, IEEE), MOUHAMMAD ALAKKOUMI[©]¹, (Member, IEEE), AND SAWSAN KANJ[©]²

¹College of Engineering and Technology, American University of the Middle East, Egaila 54200, Kuwait ²Renault Group, 92100 Boulogne-Billancourt, France

Corresponding author: Yehia Kotb (Yehia.Kotb@aum.edu.kw)

ABSTRACT Dangerous Goods Transportation in smart cities is one of the most intriguing topics that researchers are interested in to reduce the risks caused by hazardous events. This article proposes a novel technique to avoid hazardous events during the transportation of dangerous goods using real-time information provided by the cloud. This article applies Fuzzy Analytical Hierarchy Process - Technique for Order Preference by Similarity to Ideal Solution (AHP-TOPSIS) hybrid method for risk analysis applied to Dangerous goods transportation in a smart city. The article assumes that we have three criteria that dictate the choice of routs the vehicle follows: cost, duration, and risk. The main objective is to choose the best route through calculated weights assigned to the three criteria based on predefined pairwise comparisons. The study is done on both static and dynamic environments. In static environments, the decision is made and is followed prior to actions and no change of actions is made even if criteria values change. In dynamic environments on the other hand, the decision could change during the transportation process due to the change of the valuation the criteria contribute to. The study shows that risk is reduced and therefore safety is increased for the transportation of dangerous goods.

INDEX TERMS Supply chains, industry 4.0, best trajectory identification, dangerous good transportation, fuzzy AHP, fuzzy TOPSIS, multi-criteria decision making, risk evaluation.

I. INTRODUCTION

Supply chains are sets of actions and activities performed by organizations to deliver goods to customers [1]. As shown in Figure 1, a supply chain has 5 stages from suppliers to customers. Suppliers provide raw material, followed by procurement process then the production process takes place followed by the delivery process in which products are delivered to customers.

Like any big relatively new concept, Industry 4.0 does not have a unique agreed upon definition. However, most research defined it to be the result of integrating intelligence with all stages of the process [2]. In [3], industry 4.0 was

The associate editor coordinating the review of this manuscript and approving it for publication was Giovanni Pau^(D).

defined to be the process of managing the cyber-physical production network that binds together ensuring: raw material supply, delivery to customers, logistics, factory operations and manufacturing operations. Figure 2 shows the structure of industry 4.0 as described in [3].

Industry 4.0 can be seen as a whole framework or digital organization [3], [4]. This framework integrates many modern techniques such as intelligent internet of things (IIoT) [5], [6]. big data [6], [7], wireless sensor networking [8], and automation [9], [10]. It also integrate the application of artificial intelligence in supply chains [11], [12], [13], [14], [15], and the analysis of everything with the objective to improve performance while making sure that customer is satisfied [16]. Figure 3 shows the basic technological requirements of industry 4.0.





FIGURE 1. Simple supply chain model.

Supply chain 4.0 is a prerequisite for Industry 4.0. Supply chain 4.0 is a supply chain that respects the definition and constraints of industry 4.0 [4]. Supply chain 4.0 integrates intelligence with supply chain components for better planning and decision making. The objectives of applying intelligence to supply chain might change from one industry to another, however, regardless what the objectives are, Supply chain 4.0 definitely guarantees better performance in objective achievement compared to traditional supply chains [17].

As seen in Figures [1,2], delivery is an important stage in supply chains and Industry 4.0 in general. Delivering goods is not a trivial task. The process of goods delivery is a multi-criteria decision making (MCDM) process that aims to minimize *cost*, avoid *delays* and most importantly avoid *hazards* [18]. AHP [19] and TOPSIS [20] methods are the most used for MCDM problems. AHP method is especially suitable for weighing qualitative data and using pairwise comparison for each criteria [21]. The TOPSIS method has been implemented successfully to rank alternatives and take decision for different MCDM problems [22], [23], [24].

Avoiding or mitigating hazards related to DG shipment is a critical step in the delivery process since transportation presents high risk to human beings in the surrounding areas. Despite all activities of safety and security, risk exists and its level ranges from low/acceptable to high/unacceptable. This risk cannot be eliminated but it can be managed by reducing the quantity of loaded materials, changing the transport route, or the departure time [25]. Risk management involves a number of approaches, including the identification, evaluation and control of risk [26], [27], [28], [29], [30]. Generally, Risk evaluation uses two parameters: the likelihood of an incident and the severity of its outcome [31]. Those parameters depend FIGURE 2. Components of Industry 4.0.

on many inputs where some of them are uncertain. At this level, the shipment process becomes complex as it depends on three factors: cost, duration and risk level which is presented with uncertainty. The aim of this work is to propose an approach that optimizes the management of DGT delivery process. This approach uses a combination of AHP/TOPSIS methods with the integration of fuzzy models. It provides a dynamic and real-time ranking of alternative routes for the transportation process [32], [33].



FIGURE 3. Technologies of Industry 4.0.

This paper is organized as follows: Section I introduces the paper and justifies the proposed work. Section II identifies the previous work that is related to the proposed model in

this paper. Section III introduces the needed preliminaries for the reader to easily navigate through the paper. Section IV states the problem being solved in this paper and section V defines the objective of this work. Section VI specifies the contribution of this paper. Section VII presents the proposed model. Section VIII discusses the simulation and experiment results. Section IX concludes the paper.

II. LITERATURE SURVEY

In literature, many papers have been addressed the routing problem and some of them discussed the transportation of dangerous goods.

As for the risk assessment process, most of the scanned publications applied two parameters: likelihood and consequences, as effective factors that can have an effect on risk analysis [25], [34], [35], [36] and best trajectory selection for transportation [37], [38], [39]. Whereas, in order to optimize the dangerous goods transportation management (DGTM), it is important to take into account other factors than can have an effect on DGTM as trip duration and related cost. Therefore, the problem becomes a multi-criteria decision making (MCDM) problem [40], [41], [42], [43], [44]. And an appropriate method is needed for risk analysis of dangerous goods transportation to solve the interdependent issues in an effective manner.

MCDM methods have been used in several studies to analyze industrial risks [43], (e.g using the analytic network process and fuzzy linguistic approach [45], the best-worst method [46], the failure mode and effects analysis [47], [48]. Boateng et al. [49] defined a combined methodology of ANP and risk priority index to rank transportation risks.

In [50], authors focused on the mitigation of population exposure risk via production of truck-routes. To do so, they employed a single parameter metaheuristic algorithm. In [51], authors proposed a decision support system for assessing alternative distribution routes in the transportation of dangerous goods. This system considers three parameters: travel time, risk, and evacuation implications. In [52], authors discussed the problem of finding a set of non-dominated shortest paths in stochastic transportation networks using probability distribution approach. They divide the time horizon into many slots then considers a static analysis of the used parameters per time slot. In those work, authors assume that the inputs parameters are static and they didn't consider the uncertainty of the used parameters.

Many methods are proposed to solve multi-criteria decision making problems [53], [54], [55], [56]. Some MCDM methods have been used to determine the criteria weights, including the analytic hierarchy process (AHP) [57]. AHP method is especially suitable for weighting qualitative data with a crisp 9-point rating scale [58] and using pairwise comparison for each criteria [21]. It has been used in various research areas including evaluation, selection and forecasting [59], [60]. Also, AHP was used by

Zayed et al. [61] to assess risks in Chinese highway construction and in [62] to evaluate potential risk of crane incidents in construction projects. Nonetheless, AHP has a main drawback while used with a large number of items to be determined and prioritized, as it can compare only a limited number of decision alternatives [58]. Another limitation of traditional AHP is that experts cannot truly express their judgments by the crisp values in the rating scale [63]. This issue can be solved by using Fuzzy set theory.

Technique for order preference by similarity to ideal solution (TOPSIS) [22] methods are well-known methods for MCDM problems [23], [24]. TOPSIS method is used by Zavadskas et al. [64], to propose a methodology for risk assessment in construction projects.

In most of the work in risk assessment, there is a difficulty in obtaining precise information, due to: insufficient data, the number of sources of data and the vague characteristics. Uncertainty analysis has been done in several studies to represent and propagate uncertainty in risk analysis [65], [66], [67], [68], [69]. Also, the application of the fuzzy set theory may systemically improve the ability of a MCDM to handle uncertainty [34], [70], [71], [72], [73], [74], [75], [76], and [77]. Fuzzy models are helpful for demystifying, assessing and learning about risks that are not well understood. Thus, several researchers integrate fuzzy set theory with MCDM methods to handle the uncertainty and complex problems involved in a decision-making environment [78], [79], [80], [81].

Zeng et al. [82] applied Fuzzy AHP method to prioritize the risk factor in construction projects. Shin et al. [83] used AHP and Fuzzy AHP at nuclear power plant to identify and assess probable risks. Such applications show that applying Fuzzy AHP method is more appropriate for risk evaluation [43].

Some authors proposed a hybrid model that consists of both methods AHP and TOPSIS [84]. Taylan et al. [85] proposed a hybrid model for risk assessment using Fuzzy AHP and TOPSIS. Kuo and Lu [86] used Fuzzy Multiple Attribute Decision-making (MADM) approach to assess risk of a metropolitan construction project.

Risk analysis can be done using many methods such as Monte Carlo Simulation [87], [88], [89], [90], [91], Event Trees [92] and [93], Fault Trees [94], [95], and [25], Dynamic Event trees [96], Markov models [97], Failure mode and Effective Analysis [76], [98], and [99], and Petri net [100], [101], [102], and [103].

Due to the importance of these products and the critical related risk, some researches are performed in order to evaluate risks and identify best trajectory to travel as in [104], [105], and [106].

III. PRELIMINARIES

This section introduces necessary preliminaries for the understanding of this paper. Different concepts are introduced in different subsections as seen below.

A. CRISP VS FUZZY VARIABLES

Before 1965, variables were always crisp. A crisp variable is a variable that has an exact value at any point of time. In 1965, Lotify Zadah introduced the concept of fuzziness and fuzzy logic [107]. The motivation behind proposing Fuzzy logic was mentioned in one of his famous quotes:



Lotfi A. Zadeh.

Concepts like **huge**, **gigantic**, **massive**, **tiny** and so on cannot be measured by single values, instead they represent a range in which values can be described as massive or tiny. Any of those concepts includes many crisp values and they are all linguistically described by the same concept.

Figure 4 shows the fuzzy variable on the left hand side where values fade the farthest they are from the center. On the right hand side of the figure, the variable is crisp where the value is always the same regardless where we are in the area where the variable is defined.



FIGURE 4. Fuzzy variables VS. Crisp variables.

B. MULTI-CRITERIA DECISION MAKING

Enabling decision making in any given system empowers it to become an active system with added value. Multicriteria decision making (MCDM) is a process that makes use of relevant criteria through various types of methods to decide on a targeted result. MCDM methods are used in complex systems that contain expanded processes to handle uncertainty and converge onto the desired goal. The most prominent methods of MCDM are AHP, TOPSIS, ELECTRE, and Grey Theory [108]. The most commonly used methods are AHP and TOPSIS where each has been developed to accept fuzzy systems as well. These methods are called Fuzzy AHP and fuzzy TOPSIS.

C. AHP

Analytical hierarchy process (AHP) is a powerful method used to solve complex multi-criteria decision-making (MCDM) problems. In AHP, the hierarchical technique used aims to decompose a complex problem into levels of subproblems. Each level represents a set of criteria relative to each sub-problem. The top level contains the Goal or the objective. The second level represents the set of relevant criteria. If those criteria can be divided into sub-criteria, then lower levels can represent this decomposition. Thus, the last level highlights the set of alternatives as shown in figure 5.



FIGURE 5. Hierarchy problem tree.

The analysis part of AHP is based on an additive weighting process, in which several relevant attributes are represented through their relative importance. The core of AHP is the comparison of different criteria represented in quantitative way (price, weight, or area) or even in qualitative manner (feelings, preferences, or satisfaction). This comparison is based on pairwise comparison rather than sorting, voting (e.g. assigning points) or the free assignment of priorities.

The basic AHP method does not take into account the uncertainty associated with the mapping of human judgment by perception, evaluation, improvement and selection. Figure 6 illustrates the main steps of AHP:



FIGURE 6. AHP steps.

AHP can broadly be classified to be crisp AHP or Fuzzy AHP. Crisp AHP is when the scale of relative importance between every two criteria is represented by a single crisp value. This of course ignores the possibility of inaccuracy of different comparisons. Fuzzy AHP on the other hand, represents the scale of relative importance as a triangular fuzzy number which takes into consideration the possibility of representing the scale of relative importance in an uncertain way. This leads to a better decision making process since inaccuracies are being represented in the model itself.

We start below by demonstrating the crisp AHP method first then we show how the fuzzy AHP is being extended out of the crisp approach.

1) CRISP AHP

Below we show the crisp AHP in details demonstrated by examples.

• Step 1: Use a scale of relative importance like the one represented in table 1 to construct the pairwise comparison matrix for all the criteria in the levels of the hierarchy as shown in matrix A;

TABLE 1. Scale of relative importance.

Number	Description
1	Equal importance
3	Moderate Importances
5	Strong importance
7	Very Strong Importances
9	Extreme Importances
2, 4, 6, 8	Intermediate values
1/3, 1/5, 1/7, 1/9	Values for Inverse comparison

The pairwise comparison matrix A is a square matrix with a length of the number of criteria we have. If criteria is price, look, quality, and maintenance then matrix A will simply be a 4×4 square matrix. Matrix A is described as follows:

$$A = \begin{bmatrix} 1 & a_{12} & \cdots & a_{1n} \\ a_{21} & 1 & \cdots & a_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ a_{n1} & a_{n2} & & 1 \end{bmatrix}$$

Every item $a_{ij} \in A$ represents the pairwise comparison between two criteria. An example is if first criterion is speed and second criterion is price then a_{12} represents how important speed is with respect to price whereas a_{21} represents how important price is with respect to speed. This means that:

$$\forall a_{ij} \in A, \, a_{ij} = \frac{1}{a_{ji}} \tag{1}$$

This explains why the diagonal is always 1. It is because you are comparing the how important a criterion is with itself. This means that i = j and therefor

$$\forall a_{ij} \in A \| i = j, a_{ij} = 1 \tag{2}$$

from equations 1 and 2, we fill matrix *A* as follows:

$$A = \begin{bmatrix} 1 & a_{12} & \cdots & a_{1n} \\ \frac{1}{a_{12}} & 1 & \cdots & a_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ \frac{1}{a_{1n}} & \frac{1}{a_{1n}} & \cdots & 1 \end{bmatrix}$$

After matrix A is filled, we proceed with the next step.

• Step 2: After building pairwise comparison matrix *A*, weights for every criterion needs to be calculated. The first step to achieve this is to calculate the mean. The mean could either be arithmetic mean or geometric mean. Arithmetic mean is obtained by calculating the sum vector *S*. Sum vector *S* is the sum of the columns of matrix *A* and it is calculated as follows:

$$\forall s_i \in S, a_{ij} \in A, s_j = \sum_{i=0}^n a_{ij} \tag{3}$$

where *A* is an $n \times n$ matrix, $1 \le i \le n$ and $1 \le j \le n$. Given matrix *A* in III-C1, vector *S* is:

$$S = \begin{bmatrix} s_1 & s_2 & \cdots & s_n \end{bmatrix}$$

where $1 \le j \le n$ and $s_j \in S$ is calculated according to equation 3.

Now arithmetic mean is calculated according to the following matrix:

$$A' = \begin{pmatrix} \frac{1}{s_1} & \frac{a_{12}}{s_2} & \cdots & \frac{a_{1n}}{s_n} \\ \frac{a_{21}}{s_1} & \frac{1}{s_2} & \cdots & \frac{a_{2n}}{s_n} \\ \vdots & \vdots & \ddots & \vdots \\ \frac{a_{1n}}{s_1} & \frac{1}{a_{1n}} & \cdots & \frac{1}{s_n} \end{pmatrix}$$

In other words, arithmetic mean is calculated as:

$$a'_{ij} = \frac{a_{ij}}{\sum_{i=1}^{n} a_{ij}}$$
(4)

After calculating the arithmetic mean matrix A', we can now find the weight vector W where $W = \{w_1, w_2, \ldots, w_n\}$ as follows:

$$w_i = \frac{1}{n} \sum_{i=1}^{n} (a'_{ij}) \text{ for } 1 \le i \le n \text{ and } 1 \le j \le n$$
 (5)

Geometric mean is calculated as follows:

$$r_i = (a_{i1} \otimes \cdots \otimes a_{ij} \otimes \cdots \otimes a_{in})^{\frac{1}{n}}$$
(6)

where $1 \le i \le n$ and $1 \le j \le n$ and r_i is the geometric mean of row number *i*. Now the weights based on the geometric mean are calculated as:

$$\omega_i = r_i \otimes (r_1 \oplus \dots \oplus r_i \oplus \dots \oplus r_n)^{-1}$$
(7)

where $1 \le i \le n$ and $1 \le j \le n$

Note: If many sources of information are available (k experts), matrices aggregate is needed using this equation:

$$\tilde{w_{ij}} = \frac{1}{k} (\tilde{w_{ij}^1} + \tilde{w_{ij}^2} + \dots + \tilde{w_{ij}^k})$$
(8)

As seen above, after the aggregation of the opinions of the experts, the geometric means and then the local weights in each level can be determined. The results are related to each level alone and after that the global weights of the criteria that considered as final criteria is computed. This is shown in table 2.

TABLE 2. Criteria weights.

Criteria weights	w_1	w_2	 w_n	
	c1	c2	 cn	Weighted
				sum value
c1	$1 * w_1$	$a_{12} * w_2$	 $a_{1n} * w_n$	$\sum_{j=1}^{n} (a_{1j} * w_j)$
c2	$a_{21} * w_1$	$1 * w_2$	 $a_{2n} * w_n$	$\sum_{j=1}^{n} (a_{2j} * w_j)$
	$a_{n1} * w_1$	$a_{n2} * w_2$	 $1 * w_n$	$\sum_{j=1}^{n} (a_{nj} * w_j)$

• Step 3: On this step, consistency index *C.I.* is calculated. Consistency index is the most important measurement in this process and it is defined to be the index of the consistency of judgements across all pairwise comparisons.

Consistency index is defined to be a function of λ_{max} , where λ_{max} is evaluated based on the following equation:

$$\lambda_{max} = \frac{\sum_{j=1}^{n} \frac{(\text{Weighted sum value})_j}{w_j}}{n} \qquad (9)$$

After calculating $\lambda_m ax$, we can proceed to calculate the consistency index as follows:

$$C.I. = \frac{\lambda_{max} - n}{n - 1} \tag{10}$$

• Step 4: Calculate the consistency ratio *C.R.* by dividing the consistency index *C.I.* with a random index *R.I.* as shown in equation 11. Random index is the consistency index of randomly generated pairwise matrix. Table 3 illustrates the random index table for up to 10 criteria. C.R. should be less than 0.1 to be considered as reasonably consistent and to be able to move to the next step.

$$C.R. = \frac{C.I.}{R.I.} \tag{11}$$

TABLE 3. Random index table.

n	1	2	3	4	5
R.I	0.00	0.00	0.58	0.90	1.12
n	6	7	8	9	10
R.I	1.24	1.32	1.41	1.45	1.49

• Step 5: Get the global weight for each criteria. For the criteria without parent-criteria, their global and local weights are equal. For sub-criteria, their global weight is equal to their local weight multiplied by the global weight of their parent.

• Step 6: Compute the overall score of each alternative *i* with respect to all criteria as follows:

$$Score_i = \sum_{j=1}^n a_{ij} * w_j, i = 1, 2, ..., m$$
 (12)

2) FUZZY AHP

In AHP, we relied on the scale of relative importance to create the pair-wise comparison matrix for our problem. The scale contains crisp values to indicate the importance of the criteria. In fuzzy AHP, the scale is generated using triangular fuzzy numbers instead of crisp numeric values. As a result, we start this process with:

• Step 1: define the criteria matrix as follows:

$$\tilde{A} = \begin{bmatrix} 1 & a_{12} & \cdots & a_{1n} \\ a_{21} & 1 & \cdots & a_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ a_{n1} & a_{n2} & \cdots & 1 \end{bmatrix}$$

which is evaluated as follows:

$$\tilde{A} = \begin{bmatrix} 1 & a_{12} & \cdots & a_{1n} \\ \frac{1}{a_{12}} & 1 & \cdots & a_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ \frac{1}{a_{1n}} & \frac{1}{a_{2n}} & \cdots & 1 \end{bmatrix}$$

where:

$$\tilde{a}_{ij} \begin{cases} = 1, & \text{if } i = j \\ \in \left\{9^{\tilde{-}1}, \cdots, 2^{\tilde{-}1}, \tilde{2}, \cdots, \tilde{9}\right\} & \text{otherwise} \end{cases}$$
(13)

• Step 2: calculate the fuzzy geometric mean using equation 14:

$$\tilde{r}_i = (\tilde{a_{i1}} \otimes \dots \otimes \tilde{a_{ij}} \otimes \dots \otimes \tilde{a_{in}})^{1 \neq n}$$
(14)

• Step 3: calculate the fuzzy weights of each criteria using equation 15:

$$\tilde{w_i} = \tilde{r_i} \otimes (\tilde{r_1} \oplus \dots \oplus \tilde{r_i} \oplus \dots \oplus \tilde{r_n})^{-1}$$
(15)

- Step 4: change the fuzzy weights to crisp numerical value (defuzzification) using the center of area method. The method requires getting the average of the lower, middle and upper values of the triangular fuzzy number.
- Step 5: continue the AHP steps 3-6 to finally get the overall score of each alternative.

D. TOPSIS

TOPSIS is a great technique for prioritizing and selecting one or more alternatives from a pool of possible alternatives based on a set of many different criteria [109]. TOPSIS was developed by [110]. It is used to choose an alternative that has the shortest distance from the positive-ideal solution and the farthest distance from the negative-ideal solution. TOPSIS solves a problem with alternatives A_1, A_1, \dots, A_m evaluated based on n criteria $C_1, C_2, C_3, \dots, C_n$. It can be applied in two ways: Crisp and Fuzzy.

1) CRISP TOPSIS

TOPSIS method can be applied according to the following steps:

Step 1: Establish the decision matrix for ranking: in the case of m alternatives and n criteria, the decision matrix is: (X) = (x_{ij})_{m×n};

		c_1	c_2	<i>c</i> ₃		c_n
A_1	/	<i>x</i> ₁₁	<i>x</i> ₁₂	<i>x</i> ₁₃		x_{1n}
A_2		<i>x</i> ₂₁	<i>x</i> ₂₂	<i>x</i> ₂₃		<i>x</i> _{2<i>n</i>}
		÷	÷	÷	·	:
A_m	l	x_{m1}	x_{m2}	<i>x</i> _{m3}		x_{mn}

• Step 2: Normalize the decision matrix: the normalization process used in our research is based on getting the sum of each criteria (performance value) under all alternatives then divide the criteria value by that sum. For the decision matrix $X = (x_{ij})_{m \times n}$, the normalized matrix is $R = (r_{ij})_{m \times n}$, where:

$$r_{ij} = \frac{x_{ij}}{\sqrt{\sum_{i=1}^{m} (x_{ij})^2}}$$
(16)

- Step 3: Construct the weighted normalized decision matrix by multiplying the indices of the normalized matrix by the specific weight *w* of each criterion. To get the weight of each criteria, authors may use AHP method or Entropy method or even get the opinion of experts. Once the weight value is defined. The weighted normalized matrix is: $(V_{ij})_{m \times n}$ where $(v_{ij}) = w_j \times r_{ij}$. Since w_j is a crisp number, then $(v_{ij}) = w_j \times r_{ij}$ where w_j is the weight of the ith criterion for $i = 1, \dots, m$ and $j = 1, \dots, n$ and $\sum_{i=1}^{n} (w_i) = 1$.
- Step 4: Determine the positive and negative ideal solutions according to equations 17 and 18. For benefit criteria:

$$A^{+} = \left\{ v_{1}^{+}, v_{2}^{+}, \cdots, v_{n}^{+} \right\}, v_{j}^{+} = max_{i}(v_{ij})$$
(17)

For cost criteria:

$$A^{-} = \left\{ v_{1}^{-}, v_{2}^{-}, \cdots, v_{n}^{-} \right\}, v_{j}^{-} = min_{i}(v_{ij})$$
(18)

• Step 5: Determine the distance for each alternative (trajectory) from the ideal solutions by using equations 19 and 20. The distance for each alternative from positive and negative solutions can be calculated by the summation of the distances between the ratings of such alternatives under all criteria with the

ideal solutions sets.

$$d_i^+ = \sqrt{\sum_{j=1}^n d(v_{ij} - v_j^+)^2}$$
(19)

$$d_i^- = \sqrt{\sum_{j=1}^n d(v_{ij} - v_j^-)^2}$$
(20)

• Step 6: Calculate the closeness coefficient of each alternative which can be obtained using equation 21. The alternative with the highest closeness coefficient represents the best alternative.

$$CC_i = \frac{d_i^-}{d_i^- + d_i^+}$$
 $i = 1, \cdots, m.$ (21)

• Step 7: Rank the preference order. A large value of closeness coefficient CC_i^+ indicates a good performance of the alternative A_i . The best alternative is the one with the greatest relative closeness to the ideal solution.

2) FUZZY TOPSIS

Fuzzy TOPSIS method can be applied according to the following steps:

Step 1: Establish the decision matrix for ranking: in the case of m alternatives and n criteria, the decision matrix is: (X̃) = (x̃_{ij})_{m×n}, where x̃_{ij} = (a_{ij}, b_{ij}, c_{ij});

• Step 2: Normalize the decision matrix: the normalization process used in our research is based on getting the sum of each criteria (performance value) under all alternatives then divide the criteria value by that sum. For the decision matrix $\tilde{X} = (\tilde{x}_{ij})_{m \times n}$, the normalized matrix is $\tilde{R} = (\tilde{r}_{ij})_{m \times n}$, where:

$$\tilde{r}_{ij} = \frac{\tilde{x}_{ij}}{\sqrt{\sum_{i=1}^{m} (\tilde{x}_{ij})^2}}$$
(22)

• Step 3: Construct the weighted normalized decision matrix by multiplying the indices of the normalized matrix by the specific weight w of each criterion. To get the weight of each criteria, authors may use AHP method or Entropy method or even get the opinion of experts. Once the weight value is defined. The weighted normalized matrix is: $(\tilde{V}_{ij})_{m \times n}$ where $(\tilde{v}_{ij}) = w_i \times \tilde{r}_{ij}$. Since w_i is a triangular fuzzy number, then $(\tilde{v}_{ij}) =$ $(S(\tilde{w}_j, 0)) \times \tilde{r}_{ij}$ where w_j is the weight of the ith criterion, $\sum_{j=1}^{n} (S(\tilde{w_j}, 0)) = 1 \text{ and } S(\tilde{w_j}, 0) \ge 0 \text{ for } j = 1, \dots, n.$ • Step 4: Determine the positive and negative ideal solutions according to equations 23 and 24.

For benefit criteria:

$$A^{+} = \left\{ \tilde{v_{1}^{+}}, \tilde{v_{2}^{+}}, \cdots, \tilde{v_{n}^{+}} \right\}, \tilde{v_{j}^{+}} = max_{i}(\tilde{v_{ij}})$$
(23)

For cost criteria:

$$A^{-} = \left\{ \tilde{v_{1}}, \tilde{v_{2}}, \cdots, \tilde{v_{n}} \right\}, \tilde{v_{j}} = min_{i}(\tilde{v_{ij}})$$
(24)

• Step 5: Determine the distance for each alternative (trajectory) from the ideal solutions by using equations 25 and 26. The distance for each alternative from positive and negative solutions can be calculated by the summation of the distances between the ratings of such alternative under all criteria with the ideal solutions sets.

$$d_i^+ = \sqrt{\sum_{j=1}^n d(\tilde{v_{ij}} - \tilde{v_j^+})^2}$$
(25)

$$d_i^- = \sqrt{\sum_{j=1}^n d(\tilde{v_{ij}} - \tilde{v_j})^2}$$
(26)

• Step 6: Calculate the closeness coefficient of each alternative which can be obtained using equation 27. The alternative with highest closeness coefficient represents the best alternative.

$$CC_i = \frac{d_i^-}{d_i^- + d_i^+}$$
 $i = 1, \cdots, m.$ (27)

• Step 7: Rank the preference order. A large value of closeness coefficient CC_i^+ indicates a good performance of the alternative A_i . The best alternative is the one with the greatest relative closeness to the ideal solution.

E. RISK MANAGEMENT

Risk management is an applied scientific field that impacts any discipline posed by any risk. It constitutes a set of coordinated activities and procedures that focus on reducing risks with different priorities in a state or an organization [111]. The identification, assessment and prioritizing of risks are the main aspects of risk management that enable the monitoring, control and minimization of the frequency of severe and hazardous events [112]. Risk Management standards and models were created to provide stakeholders help in their decision process. These standards are: the Norm ISO 31000 and the ISO 73 guide. These systems of policies and practices are used to develop applications for risk identification, risk analysis, risk evaluation and risk treatment. The applications require elements of communication, consultation, establishing the context, monitoring and reviewing of risks [30], [113] [114], [115].

The implementation of this process starts with analyzing the risk by identifying the hazard sources or any related roots that can inflict property, people or environment calamities. To estimate the possible consequences of hazards, quotations grids and other simulation software are used in the steps after. In a similar approach, accident databases and expert judgments are used to approximate the accident occurrence probabilities. As a final step, risk analysis gives the characterization of the level of risk as well as the degree of confidence in the risk assessment.

Next, the attained risk level is compared to the decision criteria thresholds. The thresholds are derived from the context establishment step in the process. The comparison is important in studying the necessity of executing corrective measures. Risk control is defined as the process of diminishing probability of risk or severity of the damage incurred from a certain risk. Reducing the probability is considered as a prevention mechanism whereas reducing the severity is considered as a protection mechanism. Risk control can also be thought of as the group of actions taken to transfer risk or stop activities that lead to the risk.

Risk assessment represents one steps in the risk management process and it presents risk sources, vulnerable groups, and possible interventions. It enables policymakers to characterize the goals of the risk management programs and to define vulnerability reduction targets. As a general rule, risk can be presented as a function of two factors: *occurrence probability of an event* and *consequences related to it*.



FIGURE 7. Risk Assessment Protocol.

A risk assessment protocol is shown in figure 7. First, the protocol dictates proceeding with the identification and prioritization processes. The prioritization gives the rank to movements that may result in a more detailed risk analysis [116]. In some cases, more detailed techniques in risk analysis are needed to identify, evaluate and mitigate the hazards of risks that require additional evaluation. Next, the process requires qualitative and semi-quantitative risk analysis for these types of risks [117]. Qualitative risk assessment [118] approximate the unwanted consequences by identifying possible accident scenarios and attempts. Historical experience and engineering judgment are used to describe and compare risks.

In some scenarios, systems might need more details than what is offered in the qualitative approach. In these cases, semi-quantitative evaluation can be used. The quantification of consequence, likelihood, and/or risk level are part of the semi-quantitative risk analysis technique. This technique is easily recognized by the majority of the people working on DGT systems. A semi-quantitative risk assessment can be done using risk indexes or a risk ranking matrix.

Further to using the semi-quantitative method, if the analyzed risks require more analysis details then a full quantitative risk analysis method is conducted. This method makes use of defined values of indicators to do risk assessment. The assessment is used to dynamically manage risks, identify and prioritize technology needs and decision making as well as evaluate regulatory alternatives [119].

The applications of risk assessment are many and can be utilized in systems such as: insurance systems [120], building fire evacuation systems [121], online shopping transaction systems [122], and food security systems [123]. Risk level can be evaluated through two elements. The first is the likelihood of the occurrence of a defined hazard P and the second is the severity of the consequences of the incident (number of fatalities η). This is described in equation 28. The classification shows four levels: high, moderate, low, and negligible risk regions as presented in table 4.

$$RiskLevel = f(Likelihood, Severity)$$
 (28)

IV. PROBLEM STATEMENT

Shipment process is one of the key elements in any Supply Chain model. This process can be done in different modes such as: road, sea, air, or pipeline. Road shipment is the most critical mode as most accidents occur during road transportation. Any shipment of dangerous goods is described by two addresses: the source and the destination. Additional attributes are considered as the type and the quantity of goods. There are many available paths between the source and the destination. Each path is divided into several segments, where each one belongs to a specific area having its own characteristics such as: the population density and the weather conditions (see figure 8). In addition each segment has many parameters/attributes such as: number of lines, curve, slope, area, and type. Those characteristics/parameters served as inputs for the Risk assessment approach (see table 6).

The management of the delivery process focuses mainly on one traditional objective: finding the shortest length and/or the lowest cost of a route. This process is no longer valid in the case of DGT as such transportation represents serious threat to the population, the environment and properties in the traversed areas especially if any accident occurs during the transportation [124]. Those accidents may produce hazards such as fire, explosion, or toxic release that can generate severe outcomes with irreversible effects while impacting areas with high population density. It is crucial to avoid crossing such regions with consideration of the travel cost



FIGURE 8. Network structure.

and duration. Thus, the optimization of DGT should consider many criteria such as the risk level, travel cost and the travel duration. Furthermore, some of the data used in risk assessment are uncertain and time-dependent (dynamic). So, Crisp and static methods cannot be used to tackle such inputs. Hence, a fuzzy model is required to handle this uncertainty.

V. OBJECTIVE

The optimization of DGT is described as finding the best (optimal) trajectory to move dangerous goods from source to destination among a set of alternatives. It considers three main parameters which are: transport cost, transport duration, and transport risk level. As all of those are cost parameters, the objective function aims to reduce \mathbb{V} where \mathbb{V} is defined to be:

$$\mathbb{V} = \beta_1 \mathbb{C} + \beta_2 \mathbb{D} + \beta_3 \mathbb{R} \tag{29}$$

where \mathbb{C} is the cost in dollars of the trip, \mathbb{D} is the time spent in the trip, and \mathbb{R} is the risk that could be faced during the trip. Weights represent effectiveness of every factor in the total valuation \mathbb{V} . Note that the numbers written on edges between vertices represent the cost \mathbb{C} , duration \mathbb{D} and risk \mathbb{R} respectively.

VI. CONTRIBUTION

This work proposes a new dynamic approach for DGT management. This approach provides a real-time identification of the best route in dynamic environment. For the static environment, the assumption is that all the parameters have been calculated before the simulation and everything stays the same during the simulation. The proposed approach consists of several steps:

- Define a new model for risk assessment after a critical review of the available models.
- Evaluate the risk level, travel cost, and travel time related to every road segment.
- Propose a new method as a combination of AHP, TOPSIS methods and Fuzzy set.
- In the presence of an accident/ any change in the studied parameters after the departure of the truck, that may affect the ranking of the alternative trajectories, a real-time/dynamic evaluation of the best alternative will be done considering 2 different options: 1) having the ability of reconsidering the traversed paths/segments and

T/

		Severity of Harm		
Likelihood of	Catastrophic	Serious	Moderate	Negligible
occurrence	$\eta >= 100$	$10 <= \eta < 100$	$1 \le \eta < 10$	$\eta = 0$
Very Likely	High	High	High	Moderate
$P >= 10^{-1}$				
Likely	High	High	Moderate	Low
$10^{-5} \le P \le 10^{-1}$				
Unlikely	Moderate	Moderate	Low	Negligible
$10^{-9} \le P \le 10^{-5}$				
Remote	Low	Low	Negligible	Negligible
$P < 10^{-9}$				

2) elimination of the traversed path with consideration of the remaining paths only in the dynamic evaluation.

As some of the studied parameters are uncertain, Fuzzy Logic is applied as a combination of two methods Fuzzy AHP and Fuzzy TOPSIS. Fuzzy AHP is used to identify the important weights of the criteria. Then, Fuzzy TOPSIS is used to select the best alternative using the criteria weights given by the Fuzzy AHP. In addition, a new model is proposed to analyze risks and determine the characteristics of each segment. Finally, the proposed model is illustrated and tested for an expedition of Dangerous goods in France. Tests were done in both static and dynamic environments then a comparison between them was carried on. Simulations to find the best route were done considering four options:

- 1) Each criteria was considered alone in the decision
- 2) A combination of all criteria was considered
- 3) An implementation of Fuzzy AHP
- 4) An integration of Fuzzy AHP-TOPSIS

Fuzzy set was implemented according to the below detailed equations and membership.

A. FUZZY SETS

A fuzzy set is a two-dimensional mapping function which defines how a certain variable belongs to that set. The Y-axis is called the **membership** of the variable. The variable has a membership of 1 if it completely belongs to the set and it has a membership of 0 if it does not belong at all. A variable is assigned a membership μ such that $0 \le \mu \le 1$ depending on how much it belongs to that set. X-axis is the value of the variable that determines how high or low the membership of that variable to that set is. Figure 9 shows a system of three fuzzy sets: **cold, Room Temp**, and **Hot**.

In Figure 9, the membership of fuzzy set **cold** can be expressed mathematically as follows:

$$\mu(x) = \begin{cases} 1 & 0 \le x \le 17\\ \frac{21-x}{4} & 17 \le x \le 21\\ 0 & x \ge 21 \end{cases}$$



FIGURE 9. Fuzzy sets representing different temperature levels.

Using the same concept, we can find that fuzzy set **Room Temp** can be modeled as follows:

$$\mu(x) = \begin{cases} 0 & x \le 18 \\ \frac{x - 18}{3} & 18 \le x \le 21 \\ 1 & 21 \le x \le 23 \\ \frac{26 - x}{3} & 23 \le x \le 26 \\ 0 & x \ge 26 \end{cases}$$

and fuzzy set **hot** can be modelled as follows:

$$\mu(x) = \begin{cases} 0 & x \ge 21 \\ \frac{x - 23}{3} & 23 \le x \le 26 \\ 1 & x \ge 26 \end{cases}$$

If the value of x is 19 as an example, x will be both room temperature and cold. Those situations are discussed in Section VI-B.

In conclusion, a fuzzy set \tilde{X} is defined by its membership function that can be any number between zero and one. Membership of 0 means that the value does not belong to set \tilde{X} , membership of 1 means that the value completely belongs to the set under consideration, and membership between 0 and 1 determines the degree of membership. The representation of the fuzzy set \tilde{X} is:

$$\tilde{X} = \{ (x, \mu_{\tilde{x}}(x)) : x \in X, \, \mu_{\tilde{x}}(x) \in [0, 1] \} \,.$$
(30)

So, in the pair $(x, \mu_{\tilde{x}}(x))$, the first element x belongs to the classical set X, and the second element $\mu_{\tilde{x}}(x)$ belongs to the interval [0, 1], called Membership function.

B. FUZZY LOGIC

Fuzzy logic is a rule-based decision process that seeks to solve problems where the system is difficult to model and where ambiguity, or vagueness is abundant between two extremes. With fuzzy logic, propositions can be represented with degrees of truthfulness and falsehood. For example, the statement, today is sunny, might be 100% true if there are no clouds, 50% true if it's hazy and 0% true if it rains all day.

Assume that we have fuzzy sets *A* and *B*, Fuzzy logic has the following operations:

- 1) Union: $A \cup B \equiv \mu_A \lor \mu_B$
- 2) Intersection: $A \cap B \equiv \mu_A \wedge \mu_B$
- 3) Complement $\bar{A} \equiv \bar{\mu_A}$ such that $\bar{\mu_A} = 1 \mu_A$
- 4) Product $A.B \equiv \mu_{AB}$ such that $\mu_{AB} = \mu_A.\mu_B$
- 5) Sum $A+B \equiv \mu_{A+B}$ such that $\mu_{A+B} = \mu_A + \mu_B \mu_A \mu_B$
- 6) Bounded Sum: \oplus
- 7) Bounded Difference: \ominus
- 8) Bounded Product \odot

C. TRIANGULAR FUZZY NUMBER

Triangular Fuzzy Number (TFN) is the most frequent fuzzy number used. It is a fuzzy number represented with three points as follows: \tilde{X} =(a,c,b). This representation is interpreted as membership functions and holds the following conditions:

- *a* to *c* is increasing function,
- *c* to *b* is decreasing function,
- $a \leq c \leq b$,
- it is represented by a graph (see figure 10).



FIGURE 10. Triangular membership function.

The membership of the triangular fuzzy number is:

$$\mu_{(\tilde{A})}(x) = \begin{cases} \frac{x-a}{c-a} & \text{if } a \le x \le c, \\ \frac{b-x}{b-c} & \text{if } c \le x \le b, \\ 0 & \text{otherwise.} \end{cases}$$
(31)

In the literature, the simplified operations on the fuzzy values are often used when the membership functions are

represented by regular triangular trapezes [125]. For example, let $\tilde{A}=(a_1, a_2, a_3)$ and $\tilde{B}=(b_1, b_2, b_3)$ be triangular fuzzy numbers. The simplified arithmetical operations on them are presented as follows: (for $a_1, a_2, a_3>0$ and $b_1, b_2, b_3>0$)

$$\tilde{A} + \tilde{B} = (a_1 + b_1, a_2 + b_2, a_3 + b_3)$$
(32)

$$\tilde{A} - \tilde{B} = (a_1 - b_3, a_2 - b_2, a_3 - b_1)$$
 (33)

$$\tilde{A} \times \tilde{B} = (a_1 \times b_1, a_2 \times b_2, a_3 \times b_3)$$
(34)

$$\tilde{A} \div \tilde{B} = (a_1/b_3, a_2/b_2, a_3/b_1)$$
 (35)

$$\tilde{A^{-1}} = (a_1, a_2, a_3)^{-1} = (1/a_3, 1/a_2, 1/a_1)$$
(36)

$$d_{(\tilde{A},\tilde{B})} = \sqrt{\frac{1}{3} \left[(a_1 - b_1)^2 + (a_2 - b_2)^2 + (a_3 - b_3)^2 \right]}$$
(37)

In fact, the result from the addition and the subtraction operations of two TFNs is also a TFN, while the result from the multiplication and the division of them is not. Thus, in this paper we used another approach to define the operation of fuzzy numbers. This approach is based on α -*cuts* representation and the arithmetic operations of intervals.

VII. PROPOSED MODEL

The risk assessment framework presented in this paper contains four phases which are:

- · Declaring and identifying the risk types and factors
- Constructing the decision matrix
- Applying the fuzzy AHP to get the weight for the criteria (pairwise comparison)
- Apply Fuzzy TOPSIS method to rank the alternatives
- Selecting of the best alternative.

The subsequent collection of algorithms exemplify the procedure. Algorithm 1 constructs the pairwise comparison matrix \mathbb{PW} . The algorithm takes a 2D array \mathbb{A} as its input. For each row *i* in this array, the first element $\mathbb{A}[i][1]$ corresponds to the index of factor 1, the second element $\mathbb{A}[i][2]$ corresponds to the index of factor 2, and the third element $\mathbb{A}[i][3]$ represents the pairwise comparison between the factor with the index $\mathbb{A}[i][1]$ and the factor with the index $\mathbb{A}[i][2]$.

The computational complexity of Algorithm 1 is $\mathbb{O}(n)$. The algorithm takes as input comparison values, constructs a square matrix \mathbb{PW} , initializes the diagonal elements to 1, and then populates the matrix by assigning $\mathbb{PW}[row][col]$ with the value stored in $\mathbb{A}[index][3]$. Additionally, it sets $\mathbb{PW}[col][row]$ to the reciprocal of the value in $\mathbb{A}[index][3]$. It's important to note that in this context, the term "row" corresponds to the value stored in $\mathbb{A}[index][1]$, and "col" refers to the value stored in $\mathbb{A}[index][2]$, as indexing starts from 1.

Algorithm 2 outlines the fundamental AHP procedure. The algorithm receives the input of the pairwise comparison matrix \mathbb{PW} . This matrix is square in nature and is generated through the relationship (cross product) between factors and themselves.

Algorithm 1 Pairwise Comparison Matrix Formation PW

Require: 2D array \mathbb{A} that represents relationships between factors as integers, length as the number of factors.

Ensure: Pairwise comparison matrix formation $\mathbb{P}\mathbb{W}$

1: $\mathbb{PW} \leftarrow \operatorname{array}[\operatorname{length}][\operatorname{length}]$ 2: for index = 1 to length (A) do row $\leftarrow \mathbb{A}[\text{ index }][1]$ 3: $col \leftarrow \mathbb{A}[index][2]$ 4: 5: if row == col then \mathbb{PW} [row][col] $\leftarrow 1$ 6: 7: else $\mathbb{PW}[\text{ row }][\text{ col }] \leftarrow \mathbb{A}[\text{ index }][3]$ 8: $\mathbb{PW}[\text{ col }][\text{ row }] \leftarrow 1/\mathbb{A}[\text{ index }][3]$ 9: 10: end if 11: end for 12: return ℙW

Algorithm 2 Analytic Hierarchy Process (AHP)

```
Require: Pairwise comparison Matrix PW
Ensure: Wights \Omega
  1: for col = 1 to width (\mathbb{PW}) do
           col-sum \leftarrow \emptyset
 2:
          for row = 1 to length (\mathbb{PW}) do
 3:
               v \leftarrow \mathbb{PW}[\text{ row }][\text{ col }]
  4:
                col-sum[col] \leftarrow col-sum[col] + v
  5:
          end for
 6:
 7: end for
     for col = 1 to width (PW) do
 8:
          for row = 1 to length (\mathbb{PW}) do
 9:
               v \leftarrow \mathbb{PW}[\text{ row }][\text{ col }]/\text{ row-sum}[\text{ col }]
10:
               \mathbb{PW}[\text{ row }][\text{ col }] \leftarrow v
11:
12:
          end for
     end for
13:
     for row = 1 to length (\mathbb{PW}) do
14:
           weights \leftarrow \emptyset
15:
          v \leftarrow 0
16:
          for col = 1 to width (PW) do
17:
               v \leftarrow v + \mathbb{PW}[\text{ row }][\text{ col }]/\text{ row-sum}[\text{ col }]
18:
          end for
19:
           weights[row] \leftarrow v/ length (\mathbb{PW})
20:
21: end for
22: return weights
```

Algorithm 2 generates the weights among various factors. It has a computational complexity of $\mathbb{O}(n^2)$. The procedure commences by calculating the sum for each column, with each column being associated with a specific factor. Following the computation of column sums, the pairwise matrix is transformed into a normalized matrix by dividing each column by its respective sum. Subsequently, the averages of each row are obtained by summing the elements in each row and then dividing that sum by the length of the row. This iterative process yields the weights for each individual factor.

After calculating factor weights using the AHP method, it is essential to verify the consistency of those weights. Algorithm 3 illustrates how this is done.

Algo	rithm 3 AHP Weight Consistency Check
Requ	ire: 1) AHP process output Weights
-	2) Pairwise comparison Matrix \mathbb{PW} .
	3) Random index vector \mathbb{R} .
Ensu	re: Consistency ratio
1: f	for $col = 1$ to width (PW) do
2:	for row = 1 to length (\mathbb{PW}) do
3:	$v \leftarrow \text{weight [col]} \times \mathbb{PW}[\text{ row][col]}$
4:	$\mathbb{PW}[\text{ row }][\text{ col }] \leftarrow v$
5:	end for
6: E	end for
7:	weighted-sum $\leftarrow 0$
8: f	for $row = 1$ to length (\mathbb{PW}) do
9:	$v \leftarrow 0$
10:	for $col = 1$ to width (\mathbb{PW}) do
11:	$v \leftarrow v + \mathbb{PW}[\text{ row }][\text{ col }]$
12:	end for
13:	weighted-sum [col] $\leftarrow v$
14: e	end for
15:)	$\iota \leftarrow 0$
16: f	for $row = 0$ to length (\mathbb{PW}) do
17:	$\lambda \leftarrow \lambda + \text{ weighted-sum [row]/ weights [row]}$
18: E	end for
19:)	$\lambda \leftarrow \lambda / \text{ length}$
20:	Consistenct Index $\leftarrow (\lambda - \text{ length })/(\text{ length } - 1)$
21:	Consistency Ratio
	Consistency Index /RI[length]
22: r	return Consistency Ratio

Algorithm 3 has a computational complexity of $\mathbb{O}(n^2)$. it takes as input the weights computed through the AHP procedure using Algorithm 2. Additionally, the algorithm is supplied with the pairwise comparison matrix \mathbb{PW} , which is generated by Algorithm 1. Furthermore, a random vector index denoted as \mathbb{RI} is provided to the algorithm. The random index vector for the initial 9 random values is precisely defined in Table 5.

TABLE 5. Random index values for the first 9 factors.

n	1	2	3	4	5	6	7	8	9
\mathbb{RI}	0	0	0.58	0.9	1.12	1.24	1.32	1.41	1.45

The initial row of Table 5 illustrates distinct values of n, signifying the count of various factors in each decision problem. In the subsequent row of Table 5, \mathbb{RI} is represented, as the random index corresponding to the quantity of these distinct factors. The information within Table 5 implies that for a scenario involving n distinct factors, the associated random index should not surpass the value of $\mathbb{RI}[n]$.

The algorithm commences by multiplying the computed weights for each row with the corresponding values within

that row. This yields a weighted pairwise comparison matrix. Subsequently, the summation of values in each row is carried out, resulting in the creation of a weighted sum vector. The consistency factor, denoted as λ , is calculated by summing all the values within the weighted sum vector. This λ represents the maximum attainable consistency factor.

The consistency index is then computed using the following equation:

Consistency Index =
$$\frac{\lambda - \text{length}(\mathbb{PW})}{\text{length}(\mathbb{PW}) - 1}$$
 (38)

Subsequently, the consistency ratio is determined through the subsequent process:

Consistency Ratio =
$$\frac{\text{Consistency Index}}{\mathbb{RI}(\text{length})}$$
 (39)

The consistency ratio serves as an indicator of the extent of deviation from the expected state. In simpler terms, a lower consistency ratio signifies a higher level of accuracy in the assigned weights. Conventionally, a typical consistency ratio falls below the threshold of 0.1, which serves as the standard benchmark. Thus, if the consistency ratio measures less than 0.1, it provides the green light to proceed with the decision-making process, employing the calculated weights.

Algorithm 4 executes the initial phase of the TOPSIS (Technique for Order of Preference by Similarity to Ideal Solution) process. This algorithm takes as input the AHP (Analytic Hierarchy Process) weights obtained from algorithm 2, the decision matrix denoted as \mathbb{D} , and the optimal decision vector represented by \mathbb{B} .

The decision matrix \mathbb{D} is structured with rows corresponding to the various potential solutions and columns corresponding to the factors influencing the decision-making procedure. The optimal decision vector \mathbb{B} is a vector with a length equivalent to the number of columns in the decision matrix.

Within algorithm 4, the information conveyed by \mathbb{B} is pivotal. It guides the algorithm in determining whether the most desirable value for each factor is the maximum or minimum. In the scenario where $\mathbb{B}[1] = 1$, it signifies that selecting the maximum value for the first column in the decision matrix \mathbb{D} is the superior choice. Conversely, the optimal selection for the solution would involve the minimum value of column 1 if $\mathbb{B}[1]$ holds any other value.

The best factor value is calculated according to the following equation:

$$\mathbb{I}_{b}[\operatorname{col}] = \begin{cases} \max(\hat{\mathbb{D}}[\operatorname{col}]) & \text{if } \mathbb{B} = 1, \\ \\ \min(\hat{\mathbb{D}}[\operatorname{col}]) & \text{if } \mathbb{B} = 0. \end{cases}$$
(40)

The worst factor value is calculated according to the following equation:

$$\mathbb{I}_{w}[\operatorname{col}] = \begin{cases} \min(\hat{\mathbb{D}}[\operatorname{col}]) & \text{if } \mathbb{B} = 1, \\ \\ \max(\hat{\mathbb{D}}[\operatorname{col}]) & \text{if } \mathbb{B} = 0. \end{cases}$$
(41)

Algorithm 4 Topsis Preprocessing Require: 1) AHP weights

2) Decision matrix \mathbb{D} . 3) Best direction vector \mathbb{B} **Ensure:** Weighted Decision Matrix \mathbb{D} 1: Mean vector $\mathbb{U} \leftarrow \emptyset$ 2: for col = 1 to width (D) do 3: $\mathbb{U}[\operatorname{col}] \leftarrow 0$ 4: for row = 1 to length (\mathbb{D}) do $\mathbb{U}[\operatorname{col}] \leftarrow \mathbb{U}[\operatorname{col}] + (\mathbb{D}[\operatorname{row}][\operatorname{col}])^2$ 5: end for 6: $\mathbb{U}[\operatorname{col}] \leftarrow \sqrt{\mathbb{U}[\operatorname{col}]}$ 7: 8: end for 9: for col = 1 to width (\mathbb{D}) do for row = 1 to length (\mathbb{D}) do 10: $v \leftarrow weight[col]/\mathbb{U}[col]$ 11: $\hat{\mathbb{D}}$ [row][col] $\leftarrow \mathbb{D}$ [row][col] $\times v$ 12: 13: end for 14: end for Ideal Best Vector $\mathbb{I}_b \leftarrow \{\emptyset\}$ 15: Ideal Worst Vector $\mathbb{I}_{w} \leftarrow \{\emptyset\}$ 16: for col = 1 to width $(\hat{\mathbb{D}})$ do 17: $max \leftarrow -\infty$ 18: 19: $min \leftarrow \infty$ 20: for row = 1 to length $(\hat{\mathbb{D}})$ do if $max < \mathbb{D}[row][col]$ then 21: $max \leftarrow \hat{\mathbb{D}}[row][col]$ 22: end if 23: if $min > \hat{\mathbb{D}}[row][col]$ then 24: 25: $min \leftarrow \hat{\mathbb{D}}[row][col]$ end if 26: 27: end for if $\mathbb{B}[\operatorname{col}] == 1$ then 28: $\mathbb{I}_b[\operatorname{col}] \leftarrow max$ 29: 30: $\mathbb{I}_{w}[\operatorname{col}] \leftarrow \min$ else 31: $\mathbb{I}_b[\operatorname{col}] \leftarrow \min$ 32. $\mathbb{I}_{w}[\operatorname{col}] \leftarrow max$ 33. end if 34. 35: end for 36: return $\hat{\mathbb{D}}$, \mathbb{I}_h , \mathbb{I}_w

Algorithm 4 commences by computing the mean value for each column within the decision matrix \mathbb{D} and storing these averages in the vector \mathbb{U} . The dimensions of vector \mathbb{U} align with the number of columns present in \mathbb{D} . Subsequently, the square root of each element in \mathbb{U} is computed and recorded within the same vector \mathbb{U} . The computational complexity of Algorithm 4 is $\mathbb{O}(n^2)$.

Following this, each value within the decision matrix \mathbb{D} undergoes a transformation. The transformation involves multiplying each value by its corresponding AHP weight, divided by the square root of the associated column average. The determination of the best and worst potential candidates is achieved by analyzing the maximum or minimum values

in each column, guided by the optimal direction indicated in vector $\ensuremath{\mathbb{B}}.$

The outcomes provided by this algorithm consist of the normalized decision matrix denoted as $\hat{\mathbb{D}}$, as well as the vectors representing the best solution (\mathbb{I}_b) and the worst solution (\mathbb{I}_w).

Algorithm 5 has a computational complexity of $\mathbb{O}(n^2)$ and it represents a method centered around identifying a solution that minimizes the Euclidean distance from an ideal solution. The algorithm is formally named "Technique for Order Preference by Similarity to Ideal Solution," often referred to as Topsis. The input of algorithm 5 is the output of algorithm 4.

Algorithm 5 Topsis Process

1) Weighted decision making matrix $\hat{\mathbb{D}}$ Require: 2) Ideal best vector \mathbb{I}_b 3) Ideal worst vector \mathbb{I}_{w} **Ensure:** Best Choice 1: $\Delta \mathbb{I}_h \leftarrow \{\emptyset\}$ 2: $\Delta \mathbb{I}_w \leftarrow \{\emptyset\}$ 3: for row = 1 to length $(\hat{\mathbb{D}})$ do 4: $v_h \leftarrow 0$ 5: $v_w \leftarrow 0$ for col = 1 to width $(\hat{\mathbb{D}})$ do 6: $v_b \leftarrow v_b + (\mathbb{I}_b[\text{ col }] - \hat{\mathbb{D}}[\text{ row }][\text{ col }])^2$ 7: $v_w \leftarrow v_w + (\mathbb{I}_w[\text{ col }] - \hat{\mathbb{D}}[\text{ row }][\text{ col }])^2$ 8: end for 9: $\Delta \mathbb{I}_b$ [row] $\leftarrow \sqrt{v_b}$ 10: $\Delta \mathbb{I}_{w}$ [row] $\leftarrow \sqrt{v_{w}}$ 11: 12: end for 13: Choices $\leftarrow \{\emptyset\}$ for row = 1 to length ($\Delta \mathbb{I}_b$ do 14: $v_b \leftarrow \Delta \mathbb{I}_b[\text{ row }]$ 15: $v_w \leftarrow \Delta \mathbb{I}_w[\text{ row }]$ 16: Choices [row] $\leftarrow \frac{v_w}{v_h - v_w}$ 17: 18: end for Best Choice Index $\leftarrow 1$ 19. for row = 2 to length ($\Delta \mathbb{I}_b$ do 20: **if** Choices [Best Choice Index] < Choices [row] 21: then Best Choice Index \leftarrow row 22: 23: end if 24: end for 25: Best Choice \leftarrow Choices [Best Choice Index] 26: return Best Choice

Algorithm 5 commences by initializing the sets for the closest and furthest Euclidean distance vectors ($\Delta \mathbb{I}_b$ and $\Delta \mathbb{I}_w$ respectively) as empty sets \emptyset . Subsequently, the Euclidean distance is computed for each item within a column, considering both the column's best and worst values. The vectors $\Delta \mathbb{I}_b$ and $\Delta \mathbb{I}_w$ are then updated to encompass the sum square error values for both the best and worst scenarios. The performance measure is subsequently computed for each

option using the following equation:

performance[row] =
$$\frac{\Delta \mathbb{I}_{w}[row]}{\Delta \mathbb{I}_{b}[row] + \Delta \mathbb{I}_{w}[row]}$$
 (42)

The algorithm ends by returning the choice with the highest calculated performance.

The processes is detailed in the following subsections:

A. RISK ASSESSMENT PROCESS INITIALIZATION:

Risk factors are studied in risk assessment to show their significance and effect on the process. Some of the factors are attributes of risk amplification and mitigation, hazard attributes, or vulnerabilities attributes [126]. The latter being the attributes that have a direct impact on the population and the environment.

In our proposed method, to achieve the best possible route, the following categories are studied: 1) Travel Time, 2) Travel Cost and 3) Risk Level. The travel time and cost are analyzed in the next section. Risk Level is determined by studying the probability of the risk as well as its severity. The probability of the risk is affected by three factors (weather conditions, traffic density, and road characteristics). On the other hand, the severity of the incident is affected by the population density at the location of the incident.

Below is the description of the stated factors:

- Severity Characteristics: is measured by human severity and environment severity. The human severity can be determined according to the number of dead people, the number of injured people from the incident and the probability of occurrence of this incident.
- Traffic Condition: is described as Low intensity, Medium intensity, and High intensity.
- Road Characteristics: are represented by road slope, road curve, and road type.
- Weather Conditions: are categorized as Fine, Rain/Fog, and Snow/Ice

Random values of the discussed factors are introduced to show the related calculation of the process as shown in table 6. A discussion of this calculation is presented in section VIII to validate the efficacy of the proposed model.

B. CONSTRUCTING THE FUZZY DECISION MATRIX:

To build the fuzzy decision matrix, the following calculations are needed:

1) TRAVEL DURATION

The travel duration (TD) of a given segment *i* is calculated using equation 43. As shown in the equation, the weather condition (h_3) and the traffic characteristics (h_4) as well as the travelled distance determine the required time to go through the segment.

$$TD_i = \frac{length_i \times h_3 \times h_4}{meanspeed}$$
(43)

where $length_i$ is the distance between the start point and the end point of segment *i*. Based on this, the total travel duration

Factor	Details	Value
Road Curve	straight road (raduis 0)	1
h_1	curved road (>200 m)	1.3
	tightly curved road(<200 m)	2.2
Road Slope	plane road (gradient g=0%)	1
h_2	ascending road (g<5%)	1.1
	steeply ascending road (g>5%)	1.2
	descending road (g<5%)	1.3
	steeply descending road (g>5%)	1.5
Road Type	tunnel	0.8
h_5	bridge	1.2
Weather	fine weather	(0.7,1,1.3)
Conditions	rain/fog	(1.05,1.51.95)
h_3	snow/ice	(1.75,2.5,3.25)
Traffic	low intensity < 500 vehicles/hours	(0.56,0.8,1.04)
Characteristics	medium intensity < 1250 vehicles/hour	(0.7,1,1.3)
h_4	with heavy traffic < 125 lorries per day	
	high intensity>1250 vehicles/hour	(0.98, 1.4, 1.82)
	high intensity >1250 vehicles/hour	(1.68,2.4,3.12)
	with heavy traffic >250 lorries per day	

TABLE 6. Parameters of amplification and mitigation.

can be calculated for one trajectory using the following equation:

$$TD = \sum_{i=1}^{k} TD_i \tag{44}$$

where k is the number of segments for each trajectory.

2) TRAVEL COST

For each segment i, the travel cost can be calculated using equation 45.

$$TC_{i} = (TD_{i} \times (costliter / min) \times (cost / liter)) + (TD_{i} \times (costdriver / min)) + costtoll$$
(45)

Thus, the total travel cost for one trajectory is:

$$TC = \sum_{i=1}^{k} TC_i \tag{46}$$

3) ACCIDENT FREQUENCY

The frequency of an accident on the j - esimo stretch [127] for each segment can be calculated using equation 47:

$$f_j = \gamma_j (\prod_{i=1}^5 h_i) \times L_j \times n_j$$
(47)

where:

 γ_j = frequency expected on the *j* – *esimo* stretch of road [accidents km-1 per vehicle] [128]

 L_i = road length [km]

 n_i = number of vehicles [vehicles]

- γ_0 = basic frequency [accidents km-1 per vehicle]
- h_i = parameters of amplification / local mitigation

40464

We use equation 48 to calculate the frequency of an accident on a trajectory T_r :

$$F(T_r) = f_1 + \sum_{j=2}^{k} (\prod_{i=1}^{j-1} (1 - f_i)) \times f_j$$
(48)

4) SEVERITY LEVEL (SL)

It is defined as a function of the Intensity Level (IL) and Vulnerability Level (VL):

$$SL = f(IL, VL) \tag{49}$$

• Intensity Level: The intensity levels of an event are estimated by recognizing the areas impacted by the accident. In the literature, many methods were suggested to calculate the impacted areas [124]. A straightforward approach is to draw circles around the center of the accident showing areas of lethal effect (A_1) and areas of irreversible effect (A_2) . The truck position at the time of the accident is the center of all drawn circles representing the defined areas. The radius of the area is calculated based on the amount of hazardous material discharged from the truck. The amount of dangerous goods is defined by the interval [1,u]. Thus, four hazardous areas $(A_{11}, A_{12}, A_{22}, A_{21})$ can be plotted as shown in figure 11. $(A_{11}andA_{12})$ represent the areas of lethal effects whereas $(A_{22}andA_{21})$ represent the areas of irreversible effects. The radius for each is determined by the following:

$$\begin{cases}
A_{11} : R_{11} = l^{0.425} \times 3.12 \\
A_{12} : R_{12} = u^{0.425} \times 3.12 \\
A_{22} : R_{22} = l^{0.405} \times 4.7 \\
A_{21} : R_{21} = u^{0.405} \times 4.7
\end{cases}$$
(50)

- Vulnerability Level: There are two dimensions for the severity of an accident. As previously defined, the intensity level focuses on identifying the impacted areas. The second dimension focuses on the list of human and environmental assets in the impacted areas.
 - The vulnerability level focuses on two factors:
 - Human Severity (*HS*): formed through two groups: Injured People (*IP*) and Dead People (*DP*). Let |*S*| be the number of people in a circular area *S*. Thus, the number of *DP* is calculated as in equation 51

$$\tilde{DP} = \{x \in [|A_{11}|, |A_{11}| + |A_{12} - A_{11}|]\}.$$
 (51)

Also, the number of IP is calculated as in equation 52:

$$\tilde{IP} = \{x \in [|A_{22} - A_{12}|, |A_{22} - A_{12}| + |A_{21} - A_{22}|]\}.$$
(52)

- Environment severity (*ES*): is determined based on the total number of parks and rivers in both A_{12} and A_{21} .



FIGURE 11. Four Hazardous Areas: two for lethal effects and two for irreversible effects.

5) BUILDING THE FUZZY DECISION MATRIX

To build the decision matrix and rank the given alternatives, the following is taken into consideration:

• For *Path* = 1 (after the first iteration), (*Alternative* = 1 *and Criteria* = 5), the decision matrix is a 1 × 5 matrix. This is presented in table 7.

TABLE 7. One Alternative.

	c_1	c_2	c_3	c_4	c_5
A_1	\tilde{IP}_{11}	\tilde{DP}_{12}	\tilde{ES}_{13}	\tilde{TD}_{14}	\tilde{TC}_{15}

• For Path = m (after the final iteration), the decision matrix is a $m \times 5$ matrix. This is presented in table 8.

TABLE 8. m Alternatives.

	c_1	c_2	c_3	c_4	c_5
$\overline{A_1}$	\tilde{IP}_{11}	\tilde{DP}_{12}	\tilde{ES}_{13}	\tilde{TD}_{14}	\tilde{TC}_{15}
A_2	$\tilde{I}p_{21}$	\tilde{DP}_{22}	\tilde{ES}_{23}	\tilde{TD}_{24}	\tilde{TC}_{25}
A_3	$I\tilde{P}_{31}$	\tilde{DP}_{32}	\tilde{ES}_{33}	\tilde{TD}_{34}	\tilde{TC}_{35}
:	:	:	:	:	:
A_m	\tilde{IP}_{m1}	\tilde{DP}_{m2}	\tilde{ES}_{m3}	\tilde{TD}_{m4}	\tilde{TC}_{m5}

C. APPLYING FUZZY TOPSIS TO THE DECISION MATRIX:

To find the alternative with the highest relative closeness to the ideal solution, We use fuzzy TOPSIS. It is a solid method used for multi-criteria compromise programming. Once the method is applied to the fuzzy decision matrix then the following parameters are estimated for each alternative: d_i^+ , d_i^- , and CC_i . This is shown in table 9.

TABLE 9. The Separation Measures and The Relative Closeness.

	d_i^+	d_i^-	CC_i
A_1	d_{11}	d_{12}	d_{13}
A_2	d_{21}	d_{22}	d_{23}
:	:	:	:
$\dot{A_m}$	d_{m1}	d_{m2}	d_{m3}

D. SELECTING THE BEST ALTERNATIVE:

The alternative with the highest value of relative closeness is the best alternative. The higher the closeness coefficient CC_i^+ the better the performance of the alternative A_i is. Table 10 shows the ranking of the alternatives under the proposed system.





VIII. SIMULATION AND RESULTS

This section is dedicated to illustrate and discuss the simulation results. The simulator is built with Python. The simulator is composed of four components. The map which is an indirect graph and that is because of the assumption that the cost, risk and distance in any road segment (graph edge) is the same regardless which direction you are going. The correctness of this implementation is verified by testing several maps and getting all possible paths from a point to another. The algorithm is successfully avoiding circular paths for soundness and it is getting all possible paths from one point to another in a correct way and it finds the best path based on a predefined criteria.

The AHP Module is the implementation of the fuzzy version of the algorithm shown in Section III-C. The second module implements the Topsis algorithm shown in Section III-D. The correctness of two modules are verified by comparing the output of the two modules with known examples that are considered as benchmarks for the two algorithms and the output of the two implementations matched. The two modules (AHP and Topsis) are concatenated together to build the proposed model. This model determines the output of Fuzzy AHP (the weights) that is fed to Topsis for selecting the best path. Figure 12 shows the map that is used in the simulation. The objective is to move dangerous goods from V_{01} to V_{50} . The objective function is to reduce \mathbb{V} defined in equation 29 where \mathbb{C} is the cost in dollars of the trip, \mathbb{D} is the time spent in the trip, and \mathbb{R} is the risk that could be faced during the trip. Weights represent effectiveness of every factor in the total valuation \mathbb{V} . Note that the numbers written on edges between vertices represent the cost \mathbb{C} , duration \mathbb{D} and risk \mathbb{R} respectively.

Note that the cost of a path is calculated as follows:

$$\mathbb{C}_t = \Sigma_{i=1}^n \mathbb{C}(s_i) \tag{53}$$

where s_i is a segment number $i, 1 \le i \le n$ that belongs to path p, $\mathbb{C}(s_i)$ is the cost of segment s_i and path p has a number of segments ||p|| = n.

Duration of a path is calculated as follows:

$$\mathbb{D}_t = \Sigma_{i=1}^n \mathbb{D}(s_i) \tag{54}$$

where s_i is a segment number $i, 1 \le i \le n$ that belongs to path p, $\mathbb{D}(s_i)$ is the duration of segment s_i and path p has a number of segments ||p|| = n.

Risk of a path is calculated as follows:

$$\mathbb{R}_t = MAX(\mathbb{R}(s_i)) \tag{55}$$

where \mathbb{R}_i is the risk of the whole path from source to destination, s_i is a segment number $i, 1 \le i \le n$ that belongs to path p, $\mathbb{R}(s_i)$ is the risk of segment s_i and path p has a number of segments ||p|| = n.



FIGURE 12. Map used in the simulation. Numbers on arcs represent $(\mathbb{C}, \mathbb{D} \text{ and } \mathbb{R})$ defined in equation [29].

Simulation is done for both static and dynamic environments starting from source vertex V_{01} to destination vertex V_{14} . A static environment is an environment that assumes that values of $(\mathbb{C}, \mathbb{D}, \mathbb{R})$ are fixed. In other words, the path that is being predetermined before the cargo leaves source to destination will not change since numbers do not change. In dynamic environments, on other hand, the values of the tuple $(\mathbb{C}, \mathbb{D}, \mathbb{R})$ can change in one or more path segments which means optimal paths could change by time. Path segments are edges between vertices as shown in figure 12.

This section is organized as follows: Subsection VIII-A1 presents the different path costs when the costs in dollars are the only criterion considered in the selection. It also shows the selected path with some statistics about the costs in dollars in the given map in Figure 12 when the environment is static.

Subsection VIII-A2 shows the different path duration. It also presents the selected path when the only criterion considered in the selection is the duration with some statistics about the duration in the given map in Figure 12 when the environment is static.

Subsection VIII-A3 highlights the different path risks. It also presents the selected path when the only criterion considered in the selection is the risk with some statistics about the duration in the given map in Figure 12 when the environment is static. Subsection VIII-A4 demonstrates the different path valuations. It also selects the path with minimum valuation when all criteria are considered when the environment is static.

Subsection VIII-A5 manifest the selection of the minimum path using fuzzy AHP explained earlier.

Subsection VIII-A6 reveals the selection of the minimum path among all available paths using fuzzy AHP-Topsis method.

Subsection VIII-B1 presents the different path costs when the costs in dollars are the only criterion considered in the selection. It also shows the selected path with some statistics about the costs in dollars in the given map in Figure 12 when the environment is dynamic.

Subsection VIII-B4 shows the different path duration. It also presents the selected path when the only criterion considered in the selection is the duration with some statistics about the duration in the given map in Figure 12 when the environment is dynamic.

Subsection VIII-B3 highlights the different path risks. It also presents the selected path when the only criterion considered in the selection is the risk with some statistics about the duration in the given map in Figure 12 when the environment is dynamic.

Subsection VIII-A4 demonstrates the different path valuations. It also selects the path with minimum valuation when all criteria are considered when the environment is dynamic.

Subsection VIII-B5 manifest the selection of the minimum path using fuzzy AHP explained earlier.

A. STATIC ENVIRONMENT

This subsection pertains to experiments that operate under the assumption of a static environment, where costs, duration, and risks remain constant as defined initially, that is, all graph edges maintain the cost, duration, and risk values throughout the entire experiment with no change.

The map we use to simulate our static environment is shown in figure 12.

1) COST BASED PATH SELECTION

The term cost, interchangeably referred to as expenses in this context, signifies the tangible monetary outlay incurred during the journey. This includes but not limited to gas, driver wages and any other expenses for the trip. The objective is to move from vertex V_1 to vertex V_{50} with minimum cost. The experiment is based on A algorithm. From source V_1 to destination V_{50} there are 1725 different paths each with its own total cost. The different costs are shown below in figure 13.

In Figure 13, an illustration of various route expenses is presented. As previously indicated, the expenses are denominated in dollars and encompass the comprehensive transportation outlays. These encompass not only the driver's remuneration and fuel expenses but also factors such as mandatory trip maintenance and related costs.

The path with minimum cost in dollars (expenses) from source defined by vertex V_{01} to destination defined by



FIGURE 13. Path cost.

vertex V₅₀ is:





FIGURE 14. Min,Max,Mean μ ,Standard deviation σ , and Median of route expenses.

The path with maximum cost in dollars (expenses) from source defined by vertex V_{01} to destination defined by vertex V_{50} is:

$$V_{01} \rightarrow V_{02} \rightarrow V_{08} \rightarrow V_{07} \rightarrow V_{05} \rightarrow V_{12} \rightarrow V_{06} \rightarrow V_{03} \rightarrow V_{09} \rightarrow V_{10} \rightarrow V_{11} \rightarrow V_{13} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{34} \rightarrow V_{35} \rightarrow V_{36} \rightarrow V_{37} \rightarrow V_{38} \rightarrow V_{39} \rightarrow V_{40} \rightarrow V_{46} \rightarrow V_{47} \rightarrow V_{48} \rightarrow V_{49} \rightarrow V_{50}$$
(57)

As previously indicated, the evaluation of each route encompasses the fusion of financial costs (expenditures), the duration required for travel between the starting point and the destination (duration), and the potential risks that cargo could encounter along specific routes (risk). In this experiment, the optimal route is chosen based solely on expenditures. Figure 14 presents a comparison among the minimum, maximum, and average costs of all routes. Note that the cost of a path is calculated as:

$$\mathbb{C}_{t}(P) = \int_{i=1}^{i=n} \mathbb{C}(E_{i}) \ \partial \ i, \quad iE_{i} \in P$$
(58)

where *n* is the number of edges in a path, $\mathbb{C}_t(P)$ is the expenses of path *P*, and $\mathbb{C}(E_i)$ is the expense of edge E_i such that edge E_i belongs to path *P*.

Figure 14 provides a statistical overview of the expense-focused route selection. The path with the lowest expenses amounts to 1264 units, while the one with the highest expenses reaches 3699 units. The average cost across all routes is 2372 units. The standard deviation is 440 units, and the median path registers expenses of 2428 units.

2) DURATION BASED PATH SELECTION

The concept of "duration" denotes the period of time consumed throughout the travel. The objective of the experiment is to traverse from vertex V_1 to vertex V_{50} in the least amount of time. The experiment is conducted utilizing the A^* algorithm. Along the path from source V_1 to destination V_{50} , there exist 1725 distinct routes, each characterized by its unique total duration. The varying duration are visualized in figure 15.



FIGURE 15. Path Duration.

Figure 15 provides a visual representation of diverse route duration. The route characterized by the shortest duration, originating at the vertex denoted by V_{01} and terminating at the vertex marked as V_{50} , is as follows:

$$V_{01} \rightarrow V_{04} \rightarrow V_{05} \rightarrow V_{07} \rightarrow V_{14} \rightarrow V_{15} \rightarrow V_{17} \rightarrow V_{18}$$

$$\rightarrow V_{19} \rightarrow V_{20} \rightarrow V_{21} \rightarrow V_{22} \rightarrow V_{23} \rightarrow V_{24} \rightarrow V_{50}$$

(59)

The route exhibiting the maximum duration, starting at the source vertex identified as V_{01} and concluding at the destination vertex denoted by V_{50} , can be described as follows:

$$V_{01} \rightarrow V_{03} \rightarrow V_{09} \rightarrow V_{10} \rightarrow V_{11} \rightarrow V_{13} \rightarrow V_{12} \rightarrow V_{06} \rightarrow V_{04} \rightarrow V_{05} \rightarrow V_{07} \rightarrow V_{08} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{41} \rightarrow V_{42} \rightarrow V_{43} \rightarrow V_{44} \rightarrow V_{45} \rightarrow V_{46} \rightarrow V_{47} \rightarrow V_{48} \rightarrow V_{49} \rightarrow V_{40} \rightarrow V_{50}$$

$$(60)$$

Displayed in Figure 16 is a comparative analysis involving the minimum, maximum, and average duration of all routes. It's worth noting that the computation of a path's cost adheres to the equation:

$$\mathbb{D}t(P) = \int_{i=1}^{i=n} \mathbb{D}(E_i) \ \partial \ i, \quad iE_i \in P$$
(61)



FIGURE 16. Min,Max,Mean μ ,Standard deviation σ , and Median of route expenses.

Here, *n* represents the count of edges within a given path, $\mathbb{D}_{t}(P)$ signifies the duration associated with path *P*, and $\mathbb{D}(E_{i})$ denotes the duration of edge E_{i} , which is a constituent of path *P*.

Figure 16 delivers a statistical summary of the route selection process centered on duration. The route characterized by the shortest duration encompasses 37 units, while the one marked with the longest duration extends to 110 units. The mean duration across all routes equates to 73 units. The standard deviation stands at 13.5 units, and the median path entails a duration of 70 units.

3) RISK BASED PATH SELECTION

The concept of "risk" denotes a measure of the difficulties and/or danger the cargo could face in some route. The objective of the experiment is to traverse from vertex V_1 to vertex V_{50} while avoiding risks as much as possible. The experiment is conducted utilizing the A^* algorithm. Along the path from source V_1 to destination V_{50} , there exist 1725 distinct routes, each characterized by its total risk. The varying risks are visualized in figure 17.



FIGURE 17. Path Risks.

Figure 17 illustrates a range of route risks. The path exhibiting the lowest level of risk, which begins at the vertex labeled as V_{01} and concludes at the vertex identified as V_{50} ,

can be described as follows:

$$V_{01} \rightarrow V_{04} \rightarrow V_{05} \rightarrow V_{07} \rightarrow V_{14} \rightarrow V_{16} \rightarrow V_{25} \rightarrow V_{26}$$

$$\rightarrow V_{27} \rightarrow V_{28} \rightarrow V_{29} \rightarrow V_{30} \rightarrow V_{31} \rightarrow V_{32} \rightarrow V_{50}$$

(62)

The pathway demonstrating the highest degree of risk, commencing from the source vertex designated as V_{01} and culminating at the destination vertex denoted by V_{50} , can be outlined as follows:

 $V_{01} \rightarrow V_{02} \rightarrow V_{08} \rightarrow V_{07} \rightarrow V_{05} \rightarrow V_{12} \rightarrow V_{06} \rightarrow V_{03} \rightarrow V_{09} \rightarrow V_{10} \rightarrow V_{11} \rightarrow V_{13} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{34} \rightarrow V_{35} \rightarrow V_{36} \rightarrow V_{37} \rightarrow V_{38} \rightarrow V_{39} \rightarrow V_{40} \rightarrow V_{46} \rightarrow V_{47} \rightarrow V_{48} \rightarrow V_{49} \rightarrow V_{50}$ (63)



FIGURE 18. Min,Max,Mean μ ,Standard deviation σ , and Median of route expenses.

Depicted in Figure 16 is a comparative examination encompassing the least, greatest, and mean risk assessments across all routes. It is important to highlight that the computation of a route's cost is governed by the following equation:

$$\mathbb{R}t(P) = \int_{i=1}^{i=n} 2^{\mathbb{R}(E_i)} \ \partial \ i, \quad iE_i \in P$$
(64)

In this context, the variable *n* represents the count of edges within a given path. $\mathbb{R}_t(P)$ denotes the level of risk associated with path *P*, and $\mathbb{R}(E_i)$ signifies the risk attributed to edge E_i , which constitutes a part of path *P*.

Figure 18 provides a statistical overview of the route selection process based on risk assessment. The route with the lowest risk is valued at 30 units, while the one with the highest risk spans 231 units. The average risk across all routes stands at 128 units. The standard deviation is 56 units, and the median path carries a risk measurement of 206 units.

4) VALUATION BASED PATH SELECTION

In this experiment, valuation based approach is adopted to select a path. Valuation is the fusion of cost, duration and risk.

The fusion is done according to the following equation:

$$\mathbb{V} = \eta_1 \times \mathbb{C} + \eta_2 \times \mathbb{D} + \eta_3 \times \mathbb{R} \tag{65}$$

 η_1 , η_2 , and η_3 are scaling parameters. As seen in previous experiments, cost is in thousands, duration is mostly in tens while risk is mostly in hundreds. This makes the cost in dollar the dominant parameter for selecting the best path. This is why, scaling values is very important as a pre-processing phase before fusing those values together. The values of η_1 , η_2 , and η_3 are set according to the maximum magnitude of every factor. In this experiment, $\eta_1 0.1.\eta_2 = 1$ and $\eta_3 = 0.5$.

The objective of the experiment is to traverse from vertex V_1 to vertex V_{50} while maintaining minimum valuation \mathbb{V} . The experiment is conducted utilizing the A^* algorithm. Along the path from source V_1 to destination V_{50} , there exist 1725 distinct routes, each characterized by its total valuation. The varying valuations are visualized in figure 19.



FIGURE 19. Path Valuations.

Figure 19 illustrates a range of route valuation. The path exhibiting the minimum valuation, which begins at the vertex labeled as V_{01} and concludes at the vertex identified as V_{50} , can be described as follows:

$$V_{01} \rightarrow V_{04} \rightarrow V_{05} \rightarrow V_{07} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{41} \rightarrow V_{42}$$
$$\rightarrow V_{43} \rightarrow V_{44} \rightarrow V_{45} \rightarrow V_{46} \rightarrow V_{40} \rightarrow V_{50}$$
(66)



FIGURE 20. Min,Max,Mean μ ,Standard deviation σ , and Median of route expenses.

The pathway demonstrating the maximum valuation, commencing from the source vertex designated as V_{01} and

culminating at the destination vertex denoted by V_{50} , can be outlined as follows:

$$V_{01} \rightarrow V_{02} \rightarrow V_{08} \rightarrow V_{07} \rightarrow V_{05} \rightarrow V_{12} \rightarrow V_{06} \rightarrow V_{03} \rightarrow V_{09} \rightarrow V_{10} \rightarrow V_{11} \rightarrow V_{13} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{34} \rightarrow V_{35} \rightarrow V_{36} \rightarrow V_{37} \rightarrow V_{38} \rightarrow V_{39} \rightarrow V_{40} \rightarrow V_{46} \rightarrow V_{47} \rightarrow V_{48} \rightarrow V_{49} \rightarrow V_{50}$$
(67)

Depicted in Figure 20 is a comparative examination encompassing the least, greatest, and mean risk assessments across all routes.

Figure 20 provides a statistical overview of the route selection process based on valuation assessment. The route with the lowest valuation is valued at 193 units, while the one with the highest valuation spans 581 units. The average valuation across all routes stands at 581 units. The standard deviation is 74 units, and the median path carries a risk measurement of 416 units.

5) AHP BASED PATH SELECTION

In subsection VIII-A4, the composite cost of the path was assumed to assign equal weights to different criteria. The division of the cost by a hundred dollars is solely intended for normalization purposes, rather than for assigning weights.

However, it's unrealistic to presuppose that all criteria carry the same degree of significance, and this is where the Analytic Hierarchy Process (AHP) comes into play. As previously observed, AHP computes weights based on predefined pairwise comparisons. These pairwise comparisons are illustrated in Figure 23.

In Figure 21(a), the pairwise comparison that signifies the relationship between cost in dollars \mathbb{C} and duration \mathbb{D} . In this instance, cost is assessed as twice as significant as duration. Similarly, in Figure 21(b), pertains to risk \mathbb{R} and cost in dollars \mathbb{D} , with risk being deemed three times more significant than cost. Figure 21(c) reflects the pairwise comparison between risk \mathbb{R} and duration \mathbb{D} , wherein risk is considered four times more significant than duration. Applying AHP using those pairwise values produces the different coefficient values shown in Figure 21(d). These precise pairwise comparisons are outlined in the following table: Using pairwise comparison values shown in table 11

TABLE 11. Pairwise comparisons among different criteria.

	C	\mathbb{D}	\mathbb{R}
C	1	2	$\frac{1}{3}$
\mathbb{D}	$\frac{1}{2}$	1	$\frac{1}{4}$
\mathbb{R}	3	4	1

will give us the following weights:

$$\beta_1 = 0.24, \beta_2 = 0.15, \beta_3 = 0.61$$
 (68)

Equation 65 will modify to be:

$$\mathbb{V} = \beta_1 \times \eta_1 \times \mathbb{C} + \beta_2 \times \eta_2 \times \mathbb{D} + \beta_3 \times \eta_3 \times \mathbb{R} \quad (69)$$



FIGURE 21. Sub-figure (a) shows the pairwise comparison between cost \mathbb{C} and duration \mathbb{D} . Sub-figure (b) shows the pairwise comparison between risk \mathbb{R} and cost \mathbb{C} . Sub-figure (c) shows the pairwise comparison between risk \mathbb{R} and duration \mathbb{D} . Sub-figure (d) shows values of weights β_1, β_2 , and β_3 after applying the AHP process.

The objective of the experiment is to traverse from vertex V_1 to vertex V_{50} while maintaining minimum valuation \mathbb{V} calculated through AHP. Along the path from source V_1 to destination V_{50} , there exist 1725 distinct routes, each characterized by its total valuation. The varying valuations are visualized in figure 22.



FIGURE 22. Path Valuations through AHP Process.

Figure 22 illustrates a range of route valuation. The path exhibiting the minimum valuation, which begins at the vertex labeled as V_{01} and concludes at the vertex identified as V_{50} , can be described as follows:

$$V_{01} \rightarrow V_{04} \rightarrow V_{05} \rightarrow V_{07} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{41} \rightarrow V_{42}$$
$$\rightarrow V_{43} \rightarrow V_{44} \rightarrow V_{45} \rightarrow V_{46} \rightarrow V_{40} \rightarrow V_{50}$$
(70)

The pathway demonstrating the maximum valuation, commencing from the source vertex designated as V_{01} and culminating at the destination vertex denoted by V_{50} , can be



FIGURE 23. Min,Max,Mean μ ,Standard deviation σ , and Median of route expenses.

outlined as follows:

$$V_{01} \rightarrow V_{02} \rightarrow V_{08} \rightarrow V_{07} \rightarrow V_{05} \rightarrow V_{12} \rightarrow V_{06} \rightarrow V_{03} \rightarrow V_{09} \rightarrow V_{10} \rightarrow V_{11} \rightarrow V_{13} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{34} \rightarrow V_{35} \rightarrow V_{36} \rightarrow V_{37} \rightarrow V_{38} \rightarrow V_{39} \rightarrow V_{40} \rightarrow V_{46} \rightarrow V_{47} \rightarrow V_{48} \rightarrow V_{49} \rightarrow V_{50}$$

$$(71)$$

Figure 23 provides a statistical overview of the route selection process based on valuation assessment. The route with the lowest valuation is valued at 338 units, while the one with the highest valuation spans 1049 units. The average valuation across all routes stands at 661 units. The standard deviation is 128 units, and the median path carries a risk measurement of 723 units.

6) AHP TOPSIS BASED PATH SELECTION

This experiment uses the AHP Topsis merge methodology. As mentioned earlier, the weights are calculated using AHP and then weights get fed into Topsis to select the best path.

The objective of the experiment is to traverse from vertex V_1 to vertex V_{50} while maintaining minimum valuation \mathbb{V} . Along the path from source V_1 to destination V_{50} , there exist 1725 distinct routes, each characterized by its total valuation. The varying valuations are visualized in figure 24.



FIGURE 24. Path Valuations through AHP/Topsis Process.

Figure 24 illustrates a range of route valuation. The path exhibiting the minimum valuation, which begins at the vertex

labeled as V_{01} and concludes at the vertex identified as V_{50} , can be described as follows:

$$V_{01} \rightarrow V_{04} \rightarrow V_{05} \rightarrow V_{07} \rightarrow V_{14} \rightarrow V_{33} \rightarrow V_{41} \rightarrow V_{42}$$

$$\rightarrow V_{43} \rightarrow V_{44} \rightarrow V_{45} \rightarrow V_{46} \rightarrow V_{40} \rightarrow V_{50}$$
(72)



FIGURE 25. Min,Max,Mean μ ,Standard deviation σ , and Median of route expenses.

The pathway demonstrating the maximum valuation, commencing from the source vertex designated as V_{01} and culminating at the destination vertex denoted by V_{50} , can be outlined as follows:

V_{01}	\rightarrow	$V_{02} \rightarrow$	$V_{08} \rightarrow$	$V_{07} \rightarrow$	$V_{05} \rightarrow$	$V_{12} \rightarrow$	$V_{06} \rightarrow$	$V_{05} \rightarrow$
<i>V</i> ₁₂	\rightarrow	$V_{06} \rightarrow$	$V_{03} \rightarrow$	$V_{09} \rightarrow$	$V_{10} \rightarrow$	$V_{11} \rightarrow$	$V_{13} \rightarrow$	V_{14}
	\rightarrow	$V_{33} \rightarrow$	$V_{34} \rightarrow$	$V_{35} \rightarrow$	$V_{36} \rightarrow$	$V_{37} \rightarrow$	$V_{38} \rightarrow$	$V_{39} \rightarrow$
V_{40}	\rightarrow	$V_{46} \rightarrow$	$V_{47} \rightarrow$	$V_{48} \rightarrow$	$V_{49} \rightarrow$	V_{50}		(73)

Depicted in Figure 20 is a comparative examination encompassing the least, greatest, and mean risk assessments across all routes.

Figure 20 provides a statistical overview of the route selection process based on valuation assessment. The route with the lowest valuation is valued at 193 units, while the one with the highest valuation spans 581 units. The average valuation across all routes stands at 581 units. The standard deviation is 74 units, and the median path carries a risk measurement of 416 units.

B. DYNAMIC ENVIRONMENT

In this section, we test our algorithms when the criteria change. In other words, the cost in dollars change due to the crookedness, the duration increases or decreases, or the risk changes.

The scenario of the simulator in this section is as follows: Transportation is from V_{01} to V_{50} . After the vehicle reaches V_{18} , criteria changes for the link from vertex v_{18} to vertex v_{19} from values described in Figure 12 to values shown in Figure 26 which means that the algorithm needs to find a better route from V_{18} to V_{50} .



FIGURE 26. Map used in the simulation. Numbers on arcs represent $(\mathbb{C}, \mathbb{D} \text{ and } \mathbb{R})$ defined in equation [29].

The cost of rerouting is measured in this paper as follows:

$$\mathbb{C}(V_i, V_j) = \mathbb{C}(V_i, V_k) + \mathbb{C}(V_k, V_j)$$
(74)

$$\mathbb{D}(V_i, V_j) = \mathbb{D}(V_i, V_k) + \mathbb{D}(V_k, V_j)$$
(75)

$$\mathbb{R}(V_i, V_j) = max(\mathbb{R}(V_i, V_k), \mathbb{R}(V_k, V_j))$$
(76)

$$\mathbb{V}(V_x, V_y) = \beta_1 \mathbb{C}(V_x, V_y) + \beta_2 \mathbb{D}(V_x, V_y) + \beta_3 \mathbb{R}(V_x, V_y)$$

$$\mathbb{V}(V_i, V_j) = \mathbb{V}(V_i, V_k) + \mathbb{V}(V_k, V_j)$$
(78)

where V_i is the starting vertex (source / beginning of the trip), V_j is the ending vertex (destination / ending of the trip) and V_k is a vertex in the middle the vehicle is in when valued changed.

1) COST BASED PATH SELECTION

As mentioned earlier, we have three criteria for measuring the performance, the cost in dollars \mathbb{C} , the duration in hours \mathbb{D} and the risk \mathbb{R} as a number where 1 is no risk and more than 1 is how risky the road is. IN this subsection, we perform the experiment of choosing the best path where the only parameter of choice is the cost in dollars \mathbb{C} . As mentioned earlier, the experiment assumes that a dramatic change happens fur to an unexpected accident.

Figure 27 shows for sub figures. Each showing three bars. The yellow bar is the case when no change happens in the rout from V_{01} to V_{50} . This bar is the case of the static environment measure. The red bar represents the measure when an accident happens, yet, the vehicle does not change its predefined rout. The blue bar represents the measure when the vehicle changes it route based on the decision making process.



FIGURE 27. Choosing path with minimum $\mathbb C$ in a dynamic environment.

Figure 27(a) shoes the comparison of cost in dollars between the static environment cost illustrated by the yellow bar, the cost in dollars when the environment changes but the vehicle continued in the predefined path illustrated by the red bar and the cost in dollars when the environment changes and a new path is determined illustrated by the blue bar. The illustration in 28(a) shows that continuing in the predefined path will cost more than 535% of the original cost in dollars. It also shows that taking the new rout costs around 21% of the cost in dollars that would be spent if the vehicle continues in the predefined path. Due to the change in environment, the new path costs 12% more than the cost in dollars of the original path.

Figure 27(b) shows the duration of the three different paths. Continuing in the predefined path will take more than 11 times the time taken in the same path if the environment did not change. Taking the new path takes less than 10% of the time taken if the vehicle decides to continue in the predefined path. The new path takes 6% more of the time taken if the original path is followed with no accidents.

Figure 27(c) shows the risk levels in the three different paths. Continuing in the predefined path will have more than 146% of the risk of the same path if the environment did not change. Taking the new path takes less than 85% of the risk if the vehicle decides to continue in the predefined path. The new path takes 27% more of the risk if the original path is followed with no accidents.

Figure 27(d). shows the valuation in the three different paths. Valuation is calculated using equation 69 assuming that $\beta_1 = \beta_2 = \beta_3 = \eta_1 = eta_2 = \eta_3 = 1$. Continuing in the predefined path will have more than 546% of the valuation of the same path if the environment did not change. Taking the new path has 21% of the valuation if the vehicle decides

to continue in the predefined path. The new path takes 14% more of the valuation if the original path is followed with no accidents.

2) DURATION BASED PATH SELECTION

In this subsection, we assume that the only criterion of choosing the path is the duration in hours \mathbb{D} . As mentioned earlier, the experiment assumes that a dramatic change happens due to an unexpected accident.

Figure 28 shows for sub figures. Each showing three bars. The yellow bar is the case when no change happens in the rout from V_{01} to V_{50} . This bar is the case of the static environment measure. The red bar represents the measure when an accident happens, yet, the vehicle does not change its predefined rout. The blue bar represents the measure when the vehicle changes it route based on the decision making process.



FIGURE 28. Choosing path with minimum \mathbb{D} in a dynamic environment.

Figure 28(a) shows the comparison of cost in dollars between the static environment cost illustrated by the yellow bar, the cost in dollars when the environment changes but the vehicle continued in the predefined path illustrated by the red bar and the cost in dollars when the environment changes and a new path is determined illustrated by the blue bar. The illustration in 28(a) shows that continuing in the predefined path will cost more than 432% of the original cost in dollars. It also shows that taking the new rout costs around 24% of the cost in dollars that would be spent if the vehicle continues in the predefined path. Due to the change in environment, the new path costs 6% more than the cost in dollars of the original path.

Figure 28(b) shows the duration of the three different paths. Continuing in the predefined path will take more than 14 times the time taken in the same path if the environment

IEEEAccess

did not change. Taking the new path takes less than 8% of the time taken if the vehicle decides to continue in the predefined path. The new path takes 27% more of the time taken if the original path is followed with no accidents.

Figure 28(c) shows the risk levels in the three different paths. Continuing in the predefined path will have more than 125% of the risk of the same path if the environment did not change. Taking the new path takes less than 82% of the risk if the vehicle decides to continue in the predefined path. The new path takes 3% more of the risk if the original path is followed with no accidents.

Figure 28(d). shows the valuation in the three different paths. Valuation is calculated using equation 69 assuming that $\beta_1 = \beta_2 = \beta_3 = \eta_1 = eta_2 = \eta_3 = 1$. Continuing in the predefined path will have more than 446% of the valuation of the same path if the environment did not change. Taking the new path has 23% of the valuation if the vehicle decides to continue in the predefined path. The new path takes 6% more of the valuation if the original path is followed with no accidents.

3) RISK BASED PATH SELECTION

In this subsection, we assume that the only criterion of choosing the path is the risk in units \mathbb{R} . As mentioned earlier, the experiment assumes that a dramatic change happens due to an unexpected accident.

Figure 29 shows for sub figures. Each showing three bars. The yellow bar is the case when no change happens in the rout from V_{01} to V_{50} . This bar is the case of the static environment measure. The red bar represents the measure when an accident happens, yet, the vehicle does not change its predefined rout. The blue bar represents the measure when the vehicle changes it route based on the decision making process.

Figure 29(a) shows the comparison of cost in dollars between the static environment cost illustrated by the yellow bar, the cost in dollars when the environment changes but the vehicle continued in the predefined path illustrated by the red bar and the cost in dollars when the environment changes and a new path is determined illustrated by the blue bar. The illustration in 29(a) shows that continuing in the predefined path will cost more than 419% of the original cost in dollars. It also shows that taking the new rout costs around 27% of the cost in dollars that would be spent if the vehicle continues in the predefined path. Due to the change in environment, the new path costs 7% more than the cost in dollars of the original path.

Figure 29(b) shows the duration of the three different paths. Continuing in the predefined path will take more than 11 times the time taken in the same path if the environment did not change. Taking the new path takes less than 10% of the time taken if the vehicle decides to continue in the predefined path. The new path takes 11% more of the time taken if the original path is followed with no accidents.

Figure 29(c) shows the risk levels in the three different paths. Continuing in the predefined path will have more than



FIGURE 29. Choosing path with minimum $\mathbb R$ in a dynamic environment.

146% of the risk of the same path if the environment did not change. Taking the new path takes less than 86% of the risk if the vehicle decides to continue in the predefined path. The new path takes 26% more of the risk if the original path is followed with no accidents.

Figure 29(d). shows the valuation in the three different paths. Valuation is calculated using equation 69 assuming that $\beta_1 = \beta_2 = \beta_3 = \eta_1 = eta_2 = \eta_3 = 1$. Continuing in the predefined path will have more than 436% of the valuation of the same path if the environment did not change. Taking the new path has 26% of the valuation if the vehicle decides to continue in the predefined path. The new path takes 14% more of the valuation if the original path is followed with no accidents.

4) VALUATION BASED PATH SELECTION

In this subsection, we assume that the criterion of choosing the path is multi-criteria valuation \mathbb{V} defined in equation 69. As mentioned earlier, the experiment assumes that a dramatic change happens due to an unexpected accident.

Figure 30 shows for sub figures. Each showing three bars. The yellow bar is the case when no change happens in the rout from V_{01} to V_{50} . This bar is the case of the static environment measure. The red bar represents the measure when an accident happens, yet, the vehicle does not change its predefined rout. The blue bar represents the measure when the vehicle changes it route based on the decision making process.

Figure 30(a) shows the comparison of cost in dollars between the static environment cost illustrated by the yellow bar, the cost in dollars when the environment changes but the vehicle continued in the predefined path illustrated by the red bar and the cost in dollars when the environment changes



FIGURE 30. Choosing path with minimum valuation in a dynamic environment.

and a new path is determined illustrated by the blue bar. The illustration in 30(a) shows that continuing in the predefined path will cost more than 453% of the original cost in dollars. It also shows that taking the new rout costs around 25% of the cost in dollars that would be spent if the vehicle continues in the predefined path. Due to the change in environment, the new path costs 13% more than the cost in dollars of the original path.

Figure 30(b) shows the duration of the three different paths. Continuing in the predefined path will take more than 12 times the time taken in the same path if the environment did not change. Taking the new path takes less than 9% of the time taken if the vehicle decides to continue in the predefined path. The new path takes 15% more of the time taken if the original path is followed with no accidents.

Figure 30(c) shows the risk levels in the three different paths. Continuing in the predefined path will have more than 136% of the risk of the same path if the environment did not change. Taking the new path takes less than 96% of the risk if the vehicle decides to continue in the predefined path. The new path takes 31% more of the risk if the original path is followed with no accidents.

Figure 30(d). shows the valuation in the three different paths. Valuation is calculated using equation 69 assuming that $\beta_1 = \beta_2 = \beta_3 = \eta_1 = eta_2 = \eta_3 = 1$. Continuing in the predefined path will have more than 469% of the valuation of the same path if the environment did not change. Taking the new path has 24% of the valuation if the vehicle decides to continue in the predefined path. The new path takes 13% more of the valuation if the original path is followed with no accidents.

5) AHP BASED PATH SELECTION

In this subsection, we assume that the criterion of choosing the path is the AHP based valuation \mathbb{V} . As mentioned earlier, the experiment assumes that a dramatic change happens due to an unexpected accident.

Figure 31 shows for sub figures. Each showing three bars. The yellow bar is the case when no change happens in the rout from V_{01} to V_{50} . This bar is the case of the static environment measure. The red bar represents the measure when an accident happens, yet, the vehicle does not change its predefined rout. The blue bar represents the measure when the vehicle changes it route based on the decision making process.



FIGURE 31. Choosing path with minimum AHP-based $\ensuremath{\mathbb{V}}$ in a dynamic environment.

Figure 31(a) shows the comparison of cost in dollars between the static environment cost illustrated by the yellow bar, the cost in dollars when the environment changes but the vehicle continued in the predefined path illustrated by the red bar and the cost in dollars when the environment changes and a new path is determined illustrated by the blue bar. The illustration in 32(a) shows that continuing in the predefined path will cost more than 445% of the original cost in dollars. It also shows that taking the new rout costs around 26% of the cost in dollars that would be spent if the vehicle continues in the predefined path. Due to the change in environment, the new path costs 14% more than the cost in dollars of the original path.

Figure 31(b) shows the duration of the three different paths. Continuing in the predefined path will take more than 11 times the time taken in the same path if the environment did not change. Taking the new path takes less than 8% of the time taken if the vehicle decides to continue in the predefined

path. The new path takes the same time taken if the original path is followed with no accidents.

Figure 31(c) shows the risk levels in the three different paths. Continuing in the predefined path will have more than 139% of the risk of the same path if the environment did not change. Taking the new path takes less than 96% of the risk if the vehicle decides to continue in the predefined path. The new path takes 35% more of the risk if the original path is followed with no accidents.

Figure 31(d). shows the AHP based valuation in the three different paths. Continuing in the predefined path will have more than 445% of the valuation of the same path if the environment did not change. Taking the new path has 26% of the valuation if the vehicle decides to continue in the predefined path. The new path takes 14% more of the valuation if the original path is followed with no accidents.

6) AHP-TOPSIS BASED PATH SELECTION

In this subsection, we assume that the criterion of choosing the path is the AHP-Topsis based valuation \mathbb{V} . As mentioned earlier, the experiment assumes that a dramatic change happens due to an unexpected accident.

Figure 32 shows for sub figures. Each showing three bars. The yellow bar is the case when no change happens in the rout from V_{01} to V_{50} . This bar is the case of the static environment measure. The red bar represents the measure when an accident happens, yet, the vehicle does not change its predefined rout. The blue bar represents the measure when the vehicle changes it route based on the decision making process.



FIGURE 32. Choosing path with minimum AHP-TOPSIS based $\ensuremath{\mathbb{V}}$ in a dynamic environment.

Figure 32(a) shows the comparison of cost in dollars between the static environment cost illustrated by the yellow

bar, the cost in dollars when the environment changes but the vehicle continued in the predefined path illustrated by the red bar and the cost in dollars when the environment changes and a new path is determined illustrated by the blue bar. The illustration in 32(a) shows that continuing in the predefined path will cost more than 425% of the original cost in dollars. It also shows that taking the new rout costs around 25% of the cost in dollars that would be spent if the vehicle continues in the predefined path. Due to the change in environment, the new path costs 6% more than the cost in dollars of the original path.

Figure 32(b) shows the duration of the three different paths. Continuing in the predefined path will take more than 11 times the time taken in the same path if the environment did not change. Taking the new path takes less than 8% of the time taken if the vehicle decides to continue in the predefined path. The new path takes the same time taken if the original path is followed with no accidents.

Figure 32(c) shows the risk levels in the three different paths. Continuing in the predefined path will have more than 240% of the risk of the same path if the environment did not change. Taking the new path takes less than 95% of the risk if the vehicle decides to continue in the predefined path. The new path has 138% risk of the original path if the original path is followed with no accidents.

Figure 32(d). shows the AHP based valuation in the three different paths. Continuing in the predefined path will have more than 420% of the valuation of the same path if the environment did not change. Taking the new path has 25% of the valuation if the vehicle decides to continue in the predefined path. The new path takes 6% more of the valuation if the original path is followed with no accidents.

IX. CONCLUSION

This work proposes a new dynamic model for the optimization of the transportation of Dangerous goods by road. It provides a real-time ranking of all the available paths between the two end points of the expedition (source and destination) by taking into account three parameters: the risk level, travel cost and travel duration. As some of the used parameters are uncertain, a combination of Fuzzy AHP and Fuzzy TOPSIS methods are proposed and used. The main advantage of the proposed model is that it considers any dynamic change in the used parameters during the simulation and ensures a revaluation and classification of the alternative paths to return the best path in real-time.

The model was tested on a graph containing 22 segments with more than 8 paths between origin and destination. Experimental results show that the AHP-TOPSIS model takes a decision that ensures satisfaction when pair comparisons defined by decision makers are fed to the system.

REFERENCES

[1] J. T. Mentzer, W. DeWitt, J. S. Keebler, S. Min, N. W. Nix, C. D. Smith, and Z. G. Zacharia, "Defining supply chain management," *J. Bus. Logistics*, vol. 22, no. 2, pp. 1–25, 2001.

- [2] M. Alloghani, D. Al-Jumeily, A. Hussain, A. J. Aljaaf, J. Mustafina, and E. Petrov, "Healthcare services innovations based on the state of the art technology trend industry 4.0," in *Proc. 11th Int. Conf. Develop. eSystems Eng. (DeSE)*, Sep. 2018, pp. 64–70.
- [3] H. Lasi, P. Fettke, H. G. Kemper, T. Feld, and M. Hoffmann, "Industry 4.0," Bus. Inf. Syst. Eng., vol. 6, no. 4, pp. 239–242, 2014.
- [4] K. Alicke, D. Rexhausen, and A. Seyfert, "Supply chain 4.0 in consumer goods," Mckinsey & Company, Tech. Rep., 2017, pp. 1–11.
- [5] F. Firouzi, K. Chakrabarty, and S. Nassif, *Intelligent Internet of Things: From Device to Fog and Cloud.* Cham, Switzerland: Springer, 2020.
- [6] K. Witkowski, "Internet of Things, big data, industry 4.0—Innovative solutions in logistics and supply chains management," *Proc. Eng.*, vol. 182, pp. 763–769, Jan. 2017.
- [7] S. B. Sriramoju, Introduction to Big Data: Infrastructure and Networking Considerations (A Division of Ignited Minds Edutech P Ltd). Horizon Books, 2017.
- [8] A. Forster, Introduction to Wireless Sensor Networks. Hoboken, NJ, USA: Wiley, 2016.
- [9] S. Aguirre and A. Rodriguez, "Automation of a business process using robotic process automation (RPA): A case study," in *Proc. Workshop Eng. Appl.* Cham, Switzerland: Springer, 2017, pp. 65–71.
- [10] A. Asatiani and E. Penttinen, "Turning robotic process automation into commercial success—Case OpusCapita," J. Inf. Technol. Teaching Cases, vol. 6, no. 2, pp. 67–74, Nov. 2016.
- [11] H. Min, "Artificial intelligence in supply chain management: Theory and applications," *Int. J. Logistics Res. Appl.*, vol. 13, no. 1, pp. 13–39, Feb. 2010.
- [12] G. Baryannis, S. Validi, S. Dani, and G. Antoniou, "Supply chain risk management and artificial intelligence: State of the art and future research directions," *Int. J. Prod. Res.*, vol. 57, no. 7, pp. 2179–2202, Apr. 2019.
- [13] R. Toorajipour, V. Sohrabpour, A. Nazarpour, P. Oghazi, and M. Fischl, "Artificial intelligence in supply chain management: A systematic literature review," *J. Bus. Res.*, vol. 122, pp. 502–517, Jan. 2021.
- [14] R. Dash, M. McMurtrey, C. Rebman, and U. K. Kar, "Application of artificial intelligence in automation of supply chain management," *J. Strategic Innov. Sustainability*, vol. 14, no. 3, pp. 43–53, 2019.
- [15] M. Pournader, H. Ghaderi, A. Hassanzadegan, and B. Fahimnia, "Artificial intelligence applications in supply chain management," *Int. J. Prod. Econ.*, vol. 2021, May 2021, Art. no. 108250.
- [16] H.-H. Hsu, C.-Y. Chang, and C.-H. Hsu, Big Data Analytics for Sensor-Network Collected Intelligence. San Mateo, CA, USA: Morgan Kaufmann, 2017.
- [17] D. Ivanov, A. Tsipoulanidis, and J. Schönberger, "Digital supply chain, smart operations and industry 4.0," in *Global Supply Chain and Operations Management*. Cham, Switzerland: Springer, 2019, pp. 481–526.
- [18] R. Asadi and M. Ghatee, "A rule-based decision support system in intelligent hazmat transportation system," *IEEE Trans. Intell. Transp. Syst.*, vol. 16, no. 5, pp. 2756–2764, Oct. 2015.
- [19] R. Dewi, W. Verina, D. H. Tanjung, and S. L. Rahayu, "Application of AHP method based on competence for determining the best graduate students," in *Proc. 6th Int. Conf. Cyber IT Service Manage. (CITSM)*, Aug. 2018, pp. 1–5.
- [20] Z. Zhou, G. Wu, X. Dong, Y. Xue, C. Wang, and R. Shi, "An comprehensive assessment model for the distribution network with the new type of loads based on the TOPSIS method," in *Proc. 2nd IEEE Conf. Energy Internet Energy Syst. Integr. (EI2)*, Oct. 2018, pp. 1–5.
- [21] A. Kengpol, W. Meethom, and M. Tuominen, "The development of a decision support system in multimodal transportation routing within greater Mekong sub-region countries," *Int. J. Prod. Econ.*, vol. 140, no. 2, pp. 691–701, Dec. 2012.
- [22] H. Sukiman, S. Jimmy, and L. Suryati, "Decision support system for academic administration staff achievement in STMIK IBBI using TOPSIS-HFLTS method," in *Proc. 3rd Int. Conf. Mech., Electron., Comput., Ind. Technol. (MECnIT)*, Jun. 2020, pp. 282–286.
- [23] L. Wang, Y. Ali, S. Nazir, and M. Niazi, "ISA evaluation framework for security of Internet of Health Things system using AHP-TOPSIS methods," *IEEE Access*, vol. 8, pp. 152316–152332, 2020.
- [24] R. Kumar, A. Irshad Khan, Y. B. Abushark, M. M. Alam, A. Agrawal, and R. A. Khan, "An integrated approach of fuzzy logic, AHP and TOPSIS for estimating usable-security of Web applications," *IEEE Access*, vol. 8, pp. 50944–50957, 2020.

- [25] H. Kanj and J.-M. Flaus, "An agent-based framework for mitigating hazardous materials transport risk," in *Proc. IEEE Int. Conf. Evolving Adapt. Intell. Syst. (EAIS)*, Dec. 2015, pp. 1–8.
- [26] J. Sun and D. Xiao, "Study on risk management of floating transportation and installation operation of immersed tunnel based on ShenZhong link," in *Proc. 5th Int. Conf. Transp. Inf. Saf. (ICTIS)*, Jul. 2019, pp. 266–270.
- [27] Z. Ren, K. Yang, and W. Dong, "Spatial analysis and risk assessment model research of arthritis based on risk factors: China, 2011, 2013 and 2015," *IEEE Access*, vol. 8, pp. 206406–206417, 2020.
- [28] M. D. Gavrilov and D. A. Trofimov, "Operational risks of power supply failure of the transport infrastructure facilities," in *Proc. IEEE Conf. Russian Young Researchers Electr. Electron. Eng. (EIConRus)*, Jan. 2019, pp. 1397–1399.
- [29] J. Hallikas and K. Lintukangas, "Purchasing and supply: An investigation of risk management performance," *Int. J. Prod. Econ.*, vol. 171, pp. 487–494, Jan. 2016.
- [30] Q. Aini, D. P. Utami, and T. J. Qomariah, "Applying house of risk analysis for supply chain risk mitigation (case study)," in *Proc. 7th Int. Conf. Cyber IT Service Manage. (CITSM)*, vol. 7, Nov. 2019, pp. 1–4.
- [31] H. Kanj and J.-M. Flaus, "A simulation approach for risk modeling and analysis based on multi-agents," in Proc. 25th Eur. Saf. Rel. Conf., 2015.
- [32] J. Guerrero-Ibáñez, S. Zeadally, and J. Contreras-Castillo, "Sensor technologies for intelligent transportation systems," *Sensors*, vol. 18, no. 4, p. 1212, Apr. 2018.
- [33] Y. Cai, Z. Liu, H. Wang, X. Chen, and L. Chen, "Vehicle detection by fusing part model learning and semantic scene information for complex urban surveillance," *Sensors*, vol. 18, no. 10, p. 3505, Oct. 2018.
- [34] W. Huang and Y. Zhang, "Railway dangerous goods transportation system risk assessment: An approach combining FMEA with pessimistic– optimistic fuzzy information axiom considering acceptable risk coefficient," *IEEE Trans. Rel.*, vol. 70, no. 1, pp. 371–388, Mar. 2021.
- [35] M. T. Riaz, M. Shah Jahan, K. S. Arif, and W. Haider Butt, "Risk assessment on software development using fishbone analysis," in *Proc. Int. Conf. Data Softw. Eng. (ICoDSE)*, Nov. 2019, pp. 1–6.
- [36] H. Kanj and J.-M. Flaus, "A meta model framework for risk analysis, diagnosis and simulation," in *Proc. Saf. Rel. Conf. ESREL*, 2014.
- [37] N. Tontchev and S. Ivanov, "Method for solving multiple criteria decision making (MCDM) problems and decision support system," in *Proc. 3rd Int. IEEE Conf. Intell. Syst.*, Sep. 2006, pp. 126–129.
- [38] G. Li, M. Ma, C. Liu, and Y. Shu, "Fuzzy multiple attribute decision routing in VANETs," in *Proc. IEEE Int. Conf. Commun. Syst.*, Nov. 2014, pp. 564–568.
- [39] R. B. Lobov, P. G. Kolpakhchyan, and V. A. Finochenko, "Multi-drive water supply system control based on the improved method of multiple criteria decision-making (MCDM)," in *Proc. Int. Multi-Conf. Ind. Eng. Modern Technol. (FarEastCon)*, Oct. 2019, pp. 1–3.
- [40] D. D. M. Banciu and M. C. G. Florea, "Decision support system based on MADM for urban transport management," in *Proc. 1st Int. Conf. Wireless Commun., Veh. Technol., Inf. Theory Aerosp. Electron. Syst. Technol.,* May 2009, pp. 128–130.
- [41] S. El Alaoui and B. Ramamurthy, "MARS: A multi-attribute routing and scheduling algorithm for DTN interplanetary networks," *IEEE/ACM Trans. Netw.*, vol. 28, no. 5, pp. 2065–2076, Oct. 2020.
- [42] E. H. Fouad, L. Cherkaoui, and D. Khadija, "Optimizing WSN lifetime based on a new centrality multi-criteria routing approach," in *Proc. Int. Conf. Adv. Commun. Technol. Netw. (CommNet)*, Apr. 2018, pp. 1–7.
- [43] A. Karamoozian, D. Wu, C. L. P. Chen, and C. Luo, "An approach for risk prioritization in construction projects using analytic network process and decision making trial and evaluation laboratory," *IEEE Access*, vol. 7, pp. 159842–159854, 2019.
- [44] H. Singh, J. Bhutani, S. Pandit, and D. Kumar, "Congestion aware algorithm using fuzzy logic to find an optimal routing path for IoT networks," in *Proc. Int. Conf. Comput. Intell. Knowl. Economy (ICCIKE)*, Dec. 2019, pp. 141–145.
- [45] M. Ilangkumaran, M. Karthikeyan, T. Ramachandran, M. Boopathiraja, and B. Kirubakaran, "Risk analysis and warning rate of hot environment for foundry industry using hybrid MCDM technique," *Saf. Sci.*, vol. 72, pp. 133–143, Feb. 2015.
- [46] M. Yazdi, A. Nedjati, E. Zarei, and R. Abbassi, "A reliable risk analysis approach using an extension of best-worst method based on democraticautocratic decision-making style," *J. Cleaner Prod.*, vol. 256, May 2020, Art. no. 120418.

- [47] H.-W. Lo, J. J. H. Liou, C.-N. Huang, and Y.-C. Chuang, "A novel failure mode and effect analysis model for machine tool risk analysis," *Rel. Eng. Syst. Saf.*, vol. 183, pp. 173–183, Mar. 2019.
- [48] Y. Kotb, M. Alakkoumi, and H. Kanj, "Sensor network data processing in smart cities," in *Proc. Int. Conf. Control, Autom. Diagnosis (ICCAD)*, Nov. 2021, pp. 1–6.
- [49] P. Boateng, Z. Chen, and S. O. Ogunlana, "An analytical network process model for risks prioritisation in megaprojects," *Int. J. Project Manage.*, vol. 33, no. 8, pp. 1795–1811, Nov. 2015.
- [50] C. Tarantilis and C. T. Kiranoudis, "Using the vehicle routing problem for the transportation of hazardous materials," *Oper. Res.*, vol. 1, no. 1, pp. 67–78, Jan. 2001.
- [51] K. G. Zografos and K. N. Androutsopoulos, "A decision support system for integrated hazardous materials routing and emergency response decisions," *Transp. Res. C, Emerg. Technol.*, vol. 16, no. 6, pp. 684–703, Dec. 2008.
- [52] M. Owais and A. Alshehri, "Pareto optimal path generation algorithm in stochastic transportation networks," *IEEE Access*, vol. 8, pp. 58970–58981, 2020.
- [53] R. I. Mukhamediev, R. Mustakayev, K. Yakunin, S. Kiseleva, and V. Gopejenko, "Multi-criteria spatial decision making supportsystem for renewable energy development in Kazakhstan," *IEEE Access*, vol. 7, pp. 122275–122288, 2019.
- [54] H.-C. Liu, M. Yang, M. Zhou, and G. Tian, "An integrated multi-criteria decision making approach to location planning of electric vehicle charging stations," *IEEE Trans. Intell. Transp. Syst.*, vol. 20, no. 1, pp. 362–373, Jan. 2019.
- [55] L. Liu, C. Lu, F. Xiao, R. Liu, and N. N. Xiong, "A practical, integrated multi-criteria decision-making scheme for choosing cloud services in cloud systems," *IEEE Access*, vol. 9, pp. 88391–88404, 2021.
- [56] S. Mohdiwale, M. Sahu, G. R. Sinha, and V. Bajaj, "Automated cognitive workload assessment using logical teaching learning-based optimization and PROMETHEE multi-criteria decision making approach," *IEEE Sensors J.*, vol. 20, no. 22, pp. 13629–13637, Nov. 2020.
- [57] K. Fathoni, I. Prasetyaningrum, and C. L. Hariyati, "Decision support system for choosing daycare in Surabaya city using analytical hierarchy process (AHP)," in *Proc. Int. Conf. Appl. Sci. Technol. (iCAST)*, Oct. 2018, pp. 544–550.
- [58] A. Hadi-Vencheh and A. Mohamadghasemi, "A fuzzy AHP-DEA approach for multiple criteria ABC inventory classification," *Expert Syst. Appl.*, vol. 38, no. 4, pp. 3346–3352, Apr. 2011.
- [59] A. Kengpol and S. Tuammee, "The development of a decision support framework for a quantitative risk assessment in multimodal green logistics: An empirical study," *Int. J. Prod. Res.*, vol. 54, no. 4, pp. 1020–1038, Feb. 2016.
- [60] A. Kengpol, S. Tuammee, and M. Tuominen, "The development of a framework for route selection in multimodal transportation," *Int. J. Logistics Manage.*, vol. 25, no. 3, pp. 581–610, Nov. 2014.
- [61] T. Zayed, M. Amer, and J. Pan, "Assessing risk and uncertainty inherent in Chinese highway projects using AHP," *Int. J. Project Manage.*, vol. 26, no. 4, pp. 408–419, May 2008.
- [62] G. Raviv, A. Shapira, and B. Fishbain, "AHP-based analysis of the risk potential of safety incidents: Case study of cranes in the construction industry," *Saf. Sci.*, vol. 91, pp. 298–309, Jan. 2017.
- [63] K. Kaewfak, V.-N. Huynh, V. Ammarapala, and N. Ratisoontorn, "A risk analysis based on a two-stage model of fuzzy AHP-DEA for multimodal freight transportation systems," *IEEE Access*, vol. 8, pp. 153756–153773, 2020.
- [64] E. K. Zavadskas, Z. Turskis, and J. Tamošaitiene, "Risk assessment of construction projects," *J. Civil Eng. Manage.*, vol. 16, no. 1, pp. 33–46, 2010.
- [65] Y. Jianxing, C. Haicheng, W. Shibo, and F. Haizhao, "A novel risk matrix approach based on cloud model for risk assessment under uncertainty," *IEEE Access*, vol. 9, pp. 27884–27896, 2021.
- [66] V. G. Semin, S. V. Grigoreva, T. V. Dmitrieva, and E. A. Ilyina, "A process model of risk management in the system of management of strategic sustainability of cargo motor transport enterprises," in *Proc. IEEE Conf. Quality Manage., Transp. Inf. Secur., Inf. Technol.*, Oct. 2016, pp. 172–175.
- [67] S. Guo and Z. Qi, "A fuzzy best-worst multi-criteria group decisionmaking method," *IEEE Access*, vol. 9, pp. 118941–118952, 2021.

- [68] S.-P. Wan, J. Yan, W.-C. Zou, and J.-Y. Dong, "Generalized Shapley choquet integral operator based method for interactive interval-valued hesitant fuzzy uncertain linguistic multi-criteria group decision making," *IEEE Access*, vol. 8, pp. 202194–202215, 2020.
- [69] L. Zhang, C. Zhang, W. Su, and S. Zeng, "A hesitant probabilistic fuzzy multi-criteria group decision-making framework for urban land consolidation in China," *IEEE Access*, vol. 8, pp. 182930–182942, 2020.
- [70] C. Li, Z. Wu, and N. Zhu, "Monte Carlo based risk analysis method for multi-attribute decision methods," in *Proc. Cross Strait Quad-Regional Radio Sci. Wireless Technol. Conf. (CSQRWC)*, Jul. 2019, pp. 1–3.
- [71] A. S. Albana and Y. Andrian Saputra, "Financial risk assessment for power plant investment under uncertainty using Monte Carlo simulation," in *Proc. Int. Conf. Technol. Policies Electr. Power Energy*, Oct. 2019, pp. 1–6.
- [72] Y. Shang, S. Nojavan, and K. Jermsittiparsert, "Risk and uncertainty analysis of cooling demand in multi-chiller system using downside risk constraints method," *IEEE Access*, vol. 8, pp. 104511–104517, 2020.
- [73] W. Jiang, Z. Zhang, and X. Deng, "A novel failure mode and effects analysis method based on fuzzy evidential reasoning rules," *IEEE Access*, vol. 7, pp. 113605–113615, 2019.
- [74] X. Wen, X. Xu, T.-M. Choi, and S.-H. Chung, "Optimal pricing decisions of competing air-cargo-carrier systems—Impacts of risk aversion, demand, and cost uncertainties," *IEEE Trans. Syst., Man, Cybern., Syst.*, vol. 50, no. 12, pp. 4933–4947, Dec. 2020.
- [75] S. Zhang, Y. Yu, J. Li, and Q. Zhu, "A risk-averse newsvendor model under the framework of uncertainty theory," *IEEE Access*, vol. 7, pp. 182632–182642, 2019.
- [76] H. Zheng and Y. Tang, "A novel failure mode and effects analysis model using triangular distribution-based basic probability assignment in the evidence theory," *IEEE Access*, vol. 8, pp. 66813–66827, 2020.
- [77] Y. Pan, L. Zhang, Z. Li, and L. Ding, "Improved fuzzy Bayesian networkbased risk analysis with interval-valued fuzzy sets and D–S evidence theory," *IEEE Trans. Fuzzy Syst.*, vol. 28, no. 9, pp. 2063–2077, Sep. 2020.
- [78] S. Boral, S. K. Chaturvedi, I. M. Howard, K. McKee, and V. N. A. Naikan, "An integrated approach for fuzzy failure mode and effect analysis using fuzzy AHP and fuzzy MARCOS," in *Proc. IEEE Int. Conf. Ind. Eng. Eng. Manage. (IEEM)*, Dec. 2020, pp. 395–400.
- [79] X. Li, Y. Liu, Y. Wang, and Z. Gao, "Evaluating transit operator efficiency: An enhanced DEA model with constrained fuzzy-AHP cones," *J. Traffic Transp. Eng.*, vol. 3, no. 3, pp. 215–225, Jun. 2016.
- [80] H. Ghunaim and J. Dichter, "Applying the FAHP to improve the performance evaluation reliability of software defect classifiers," *IEEE Access*, vol. 7, pp. 62794–62804, 2019.
- [81] M. Stanković, P. Gladović, and V. Popović, "Determining the importance of the criteria of traffic accessibility using fuzzy AHP and rough AHP method," *Decis. Making, Appl. Manage. Eng.*, vol. 2, no. 1, pp. 86–104, Mar. 2019.
- [82] J. Zeng, M. An, and N. J. Smith, "Application of a fuzzy based decision making methodology to construction project risk assessment," *Int. J. Project Manage.*, vol. 25, no. 6, pp. 589–600, Aug. 2007.
- [83] D.-W. Shin, Y. Shin, and G.-H. Kim, "Comparison of risk assessment for a nuclear power plant construction project based on analytic hierarchy process and fuzzy analytic hierarchy process," *J. Building Construction Planning Res.*, vol. 4, no. 3, pp. 157–171, 2016.
- [84] A. T. Gumus, "Evaluation of hazardous waste transportation firms by using a two step fuzzy-AHP and TOPSIS methodology," *Expert Syst. Appl.*, vol. 36, no. 2, pp. 4067–4074, Mar. 2009.
- [85] O. Taylan, A. O. Bafail, R. M. S. Abdulaal, and M. R. Kabli, "Construction projects selection and risk assessment by fuzzy AHP and fuzzy TOPSIS methodologies," *Appl. Soft Comput.*, vol. 17, pp. 105–116, Apr. 2014.
- [86] Y.-C. Kuo and S.-T. Lu, "Using fuzzy multiple criteria decision making approach to enhance risk assessment for metropolitan construction projects," *Int. J. Project Manage.*, vol. 31, no. 4, pp. 602–614, May 2013.
- [87] S. A. Silva, P. H. C. D. Abreu, F. R. D. Amorim, and D. F. L. Santos, "Application of Monte Carlo simulation for analysis of costs and economic risks in a banking agency," *IEEE Latin Amer. Trans.*, vol. 17, no. 3, pp. 409–417, Mar. 2019.
- [88] V. Sapalatua and H. N. Siagian, "Implementation of risk analysis using Monte Carlo simulation on electricity investment decision making: Case study: Steam and combined cycle power plant development in Indonesia by PT pembangkitan jawa-bali," in *Proc. Int. Conf. Technol. Policies Electric Power Energy*, Oct. 2019, pp. 1–5.

- [89] O. B. Tokdemir, H. Erol, and I. Dikmen, "Delay risk assessment of repetitive construction projects using line-of-balance scheduling and Monte Carlo simulation," *J. Construction Eng. Manage.*, vol. 145, no. 2, Feb. 2019, Art. no. 04018132.
- [90] L. G. G. Ospina, A. J. B. Rueda, A. F. M. Bessolo, and J. L. P. Tienda, "A methodological proposal for risk analysis in the construction of tunnels," in *Advances in Informatics and Computing in Civil and Construction Engineering*. Cham, Switzerland: Springer, 2019, pp. 815–822.
- [91] G. Wang, H. Xu, B. Pei, Y. Xue, and X. Duan, "Risk topology construction method of flight manipulation in turbulent-windshear conditions based on multivariate copula theory," *IEEE Access*, vol. 8, pp. 127801–127817, 2020.
- [92] J. Wang, B. Hu, L. Dong, J. Chang, and W.-P. Wang, "Safety precontrol of stope roof fall accidents using combined event tree and fuzzy numbers in China's underground noncoal mines," *IEEE Access*, vol. 8, pp. 177615–177622, 2020.
- [93] J. A. Vílchez, V. Espejo, and J. Casal, "Generic event trees and probabilities for the release of different types of hazardous materials," J. Loss Prevention Process Industries, vol. 24, no. 3, pp. 281–287, May 2011.
- [94] H. Kanj and P. E. Abi-Char, "A new fuzzy-TOPSIS based risk decision making framework for dangerous good transportation," in *Proc. IEEE* 21st Int. Conf. High Perform. Comput. Commun.; IEEE 17th Int. Conf. Smart City; IEEE 5th Int. Conf. Data Sci. Syst. (HPCC/SmartCity/DSS), Aug. 2019, pp. 2666–2672.
- [95] B. Gjorgiev, A. Antenucci, A. Volkanovski, and G. Sansavini, "An FTA method for the unavailability of supply in gas networks supported by physical models," *IEEE Trans. Rel.*, vol. 69, no. 2, pp. 740–753, Jun. 2020.
- [96] M. D'Onorio, T. Glingler, F. Giannetti, and G. Caruso, "Dynamic event tree analysis as a tool for risk assessment in nuclear fusion plants using RAVEN and MELCOR," *IEEE Trans. Plasma Sci.*, vol. 50, no. 11, pp. 4514–4520, Nov. 2022.
- [97] M. Yang, L. Jia, W. Xie, and T. Gao, "Research on risk assessment model of epidemic diseases in a certain region based on Markov chain and AHP," *IEEE Access*, vol. 9, pp. 75826–75839, 2021.
- [98] W.-Z. Wang, X.-W. Liu, and S.-L. Liu, "Failure mode and effect analysis for machine tool risk analysis using extended gained and lost dominance score method," *IEEE Trans. Rel.*, vol. 69, no. 3, pp. 954–967, Sep. 2020.
- [99] H. Zhang, Y. Dong, I. Palomares-Carrascosa, and H. Zhou, "Failure mode and effect analysis in a linguistic context: A consensus-based multiattribute group decision-making approach," *IEEE Trans. Rel.*, vol. 68, no. 2, pp. 566–582, Jun. 2019.
- [100] X.-Y. Li, Z.-L. Wang, Y. Xiong, and H.-C. Liu, "A novel failure mode and effect analysis approach integrating probabilistic linguistic term sets and fuzzy Petri nets," *IEEE Access*, vol. 7, pp. 54918–54928, 2019.
- [101] L. Yu, Q. Liu, R. Hua, and Y. Fu, "Risk analysis of cash on delivery payment method by social network analysis and fuzzy Petri net," *IEEE Access*, vol. 8, pp. 174160–174168, 2020.
- [102] M. P. Fanti, G. Iacobellis, and W. Ukovich, "A risk assessment framework for hazmat transportation in highways by colored Petri nets," *IEEE Trans. Syst., Man, Cybern., Syst.*, vol. 45, no. 3, pp. 485–495, Mar. 2015.
- [103] H. Kanj, Y. Kotb, and J.-M. Flaus, "A proposed Petri-Net extension to analyze risk for evolving systems using an agent model," in *Proc. Int. Conf. Control, Autom. Diagnosis (ICCAD)*, Oct. 2020, pp. 1–7.
- [104] F. Opitz, C. Mohrdieck, K. Dästner, J. J. N. Corcuera, E. Schmid, and B. V. H. Z. Roseneckh-Köhler, "Data analytics, machine learning and risk assessment for surveillance and situation awareness," in *Proc. 21st Int. Radar Symp. (IRS)*, Oct. 2020, pp. 173–178.
- [105] Z. Bahramian and M. Bagheri, "An approach for road, railway, pipeline routing problem in hazardous materials transportation using multiple criteria," in *Proc. 4th Int. Conf. Transp. Inf. Saf. (ICTIS)*, Aug. 2017, pp. 1088–1092.
- [106] M. Zhang, N. Wang, Z. He, Z. Yang, and Y. Guan, "Bi-objective vehicle routing for hazardous materials transportation with actual load dependent risks and considering the risk of each vehicle," *IEEE Trans. Eng. Manag.*, vol. 66, no. 3, pp. 429–442, Aug. 2019.
- [107] L. A. Zadeh, "Fuzzy logic," Inf. Control, vol. 8, no. 4, pp. 338–353, 1965.
- [108] M. Aruldoss, T. M. Lakshmi, and V. P. Venkatesan, "A survey on multi criteria decision making methods and its applications," *Amer. J. Inf. Syst.*, vol. 1, no. 1, pp. 31–43, 2013.

- [109] Y. Ji, G. H. Huang, and W. Sun, "Risk assessment of hydropower stations through an integrated fuzzy entropy-weight multiple criteria decision making method: A case study of the Xiangxi river," *Expert Syst. Appl.*, vol. 42, no. 12, pp. 5380–5389, Jul. 2015.
- [110] K. P. Yoon and C.-L. Hwang, *Multiple Attribute Decision Making:* An Introduction, vol. 104. Newbury Park, CA, USA: Sage, 1995.
- [111] F. Marle, C. Pointurier, and H. Jaber, "Managing a complex project using a risk-risk multiple domain matrix," *J. Mod. Project Manage.*, vol. 3, no. 2, p. 132, 2015.
- [112] H. Kanj, W. H. F. Aly, and S. Kanj, "A novel dynamic approach for risk analysis and simulation using multi-agents model," *Appl. Sci.*, vol. 12, no. 10, p. 5062, May 2022.
- [113] N. Azizi, S. J. Miah, and F. H. Masmali, "Development of an innovative framework for IT risk management," in *Proc. IEEE Asia–Pacific Conf. Comput. Sci. Data Eng. (CSDE)*, Dec. 2019, pp. 1–4.
- [114] D. T. Roberts, "Applying risk assessment at the worker level: Applications to electrical safety," *IEEE Ind. Appl. Mag.*, vol. 25, no. 1, pp. 18–24, Jan. 2019.
- [115] J. Selvik and E. Abrahamsen, "Explicit and implicit inclusion of time in the definitions of risk and reliability," in *Safety and Reliability*, vol. 40. New York, NY, USA: Taylor & Francis, 2021, pp. 9–27.
- [116] K. Singer and M. Menon, "Probabilistic risk assessment for acceptance testing," in *Proc. Annu. Rel. Maintainability Symp. (RAMS)*, Jan. 2020, pp. 1–3.
- [117] C. Wang and A. Mosleh, "Qualitative-quantitative Bayesian belief networks for reliability and risk assessment," in *Proc.-Annu. Rel. Maintainability Symp. (RAMS)*, Jan. 2010, pp. 1–5.
- [118] V. N. Burkov, I. V. Burkova, K. E. Amelina, A. D. Yu, and I. V. Goroshko, "Management of complex project risks based on qualitative assessments," in *Proc. 11th Int. Conf. 'Manag. Large-Scale Syst. Develop.'* (*MLSD*, Oct. 2018, pp. 1–3.
- [119] E. E. Hodgson, T. E. Essington, J. F. Samhouri, E. H. Allison, N. J. Bennett, A. Bostrom, A. C. Cullen, S. Kasperski, P. S. Levin, and M. R. Poe, "Integrated risk assessment for the blue economy," *Frontiers Mar. Sci.*, vol. 6, p. 609, Sep. 2019.
- [120] N. Dhieb, H. Ghazzai, H. Besbes, and Y. Massoud, "A secure AI-driven architecture for automated insurance systems: Fraud detection and risk measurement," *IEEE Access*, vol. 8, pp. 58546–58558, 2020.
- [121] G. N. Shariff, J. C. E. Yong, N. Salleh, and C. L. Siow, "Risk assessment of building fire evacuation with stochastic obstructed emergency exit," in *Proc. 4th Int. Conf. Workshops Recent Adv. Innov. Eng. (ICRAIE)*, Nov. 2019, pp. 1–5.
- [122] W. Cui, W. Xingfen, and Z. Wenying, "The research on crossborder online shopping transaction risk based on online data access," in *Proc. IEEE Int. Conf. Big Data (Big Data)*, Dec. 2019, pp. 5331–5335.
- [123] R. A. Permana, A. Y. Ridwan, F. Yulianti, and P. G. A. Kusuma, "Design of food security system monitoring and risk mitigation of rice distribution in Indonesia bureau of logistics," in *Proc. IEEE* 13th Int. Conf. Telecommun. Syst., Services, Appl. (TSSA), Oct. 2019, pp. 249–254.
- [124] H. Kanj, "Contribution to risk analysis related to the transport of hazardous materials by agent-based simulation," Ph.D. dissertation, Université Grenoble Alpes, 2016.
- [125] H. Jaber, F. Marle, L.-A. Vidal, I. Sarigol, and L. Didiez, "A framework to evaluate project complexity using the fuzzy TOPSIS method," *Sustainability*, vol. 13, no. 6, p. 3020, Mar. 2021.
- [126] Y. Kotb, M. Alakkoummi, and H. Kanj, "Reinforcement learning based framework for real time fault tolerance," in *Proc. 11th IEEE Annu. Inf. Technol., Electron. Mobile Commun. Conf. (IEMCON)*, Nov. 2020, pp. 0357–0364.
- [127] B. Fabiano, F. Currò, E. Palazzi, and R. Pastorino, "A framework for risk assessment and decision-making strategies in dangerous good transportation," *J. Hazardous Mater.*, vol. 93, no. 1, pp. 1–15, Jul. 2002.
- [128] D. W. Harwood, J. G. Viner, and E. R. Russell, "Procedure for developing truck accident and release rates for hazmat routing," *J. Transp. Eng.*, vol. 119, no. 2, pp. 189–199, Mar. 1993.



HASSAN KANJ (Member, IEEE) received the M.Sc. degree in software engineering from Lebanese University (LU), in 2011, the degree in computer engineering from the University of Technology of Compiégne (UTC), in 2013, and the Ph.D. degree in computer engineering from the University of Grenoble Alpes (UGA), France. In 2016, he defended his dissertation entitled "Contribution to Risk Analysis Related to the Transport of Hazardous Materials by Agent-Based

Simulation." Since 2017, he has been an Assistant Professor of computer engineering with the American University of the Middle East, Kuwait. His research interests include artificial intelligence, multi-agents modeling and simulation, and risk analysis.



MOUHAMMAD ALAKKOUMI (Member, IEEE) received the B.S. degree in computer and communications engineering from the American University of Technology, Halat, Lebanon, in 2005, and the M.S. degree in telecommunications systems and the Ph.D. degree in electrical and computer engineering from the University of Oklahoma, Norman, OK, USA, in 2007 and 2011, respectively. He joined the American University of the Middle East, in 2017. His research interests

include sensor networks, intelligent transportation systems, image processing, and optical communications.



YEHIA KOTB (Member, IEEE) received the B.Eng. and Master of Science degrees in computer engineering from Arab Academy for Science and Technology and Maritime Transport, Alexandria, Egypt, in 1996 and 2001, respectively, and the Doctor of Philosophy degree in computer science from Western University, London, ON, Canada, in 2011. From 2003 to 2005, he was a Research Assistant with the University of Mannheim, Mannheim, Germany. From 2005 to 2009, he was a

Research Assistant/a Teaching Assistant with Western University. He joined industry and worked in reputable world leading companies, such as Pivotal Labs (VMware), Toronto, ON; Autodata, London; and Phoenix Interactive (Diebold), London. He is currently an Associate Professor with the American University of the Middle East, Kuwait. His research interests include process and workflow development, management, and optimization.



SAWSAN KANJ received the Diploma degree in computer engineering from Lebanese University, and the Ph.D. degree in machine learning from the University of Technology of Compiégne, in 2013. She was an Engineer Researcher with the French Alternative Energies and Atomic Energy Commission, Paris, before joining the Renault Group, in 2018. Her research interests include machine learning, big data analytics, and multiagent systems.